DOI 10.15622/ia.22.6.1

M.J. DIVÁN, D. SHCHEMELININ, M.E. CARRANZA, C.I. MARTINEZ-SPESSOT, M. BUINEVICH REAL-TIME RELIABILITY MONITORING ON EDGE COMPUTING: A SYSTEMATIC MAPPING

Diván M.J., Shchemelinin D., Carranza M.E., Martinez-Spessot C.I., Buinevich M. Real-Time Reliability Monitoring on Edge Computing: a Systematic Mapping.

Abstract. Scenario: System reliability monitoring focuses on determining the level at which the system works as expected (under certain conditions and over time) based on requirements. The edge computing environment is heterogeneous and distributed. It may lack central control due to the scope, number, and volume of stakeholders. Objective: To identify and characterize the Real-time System Reliability Monitoring strategies that have considered Artificial Intelligence models for supporting decision-making processes. Methodology: An analysis based on the Systematic Mapping Study was performed on December 14, 2022. The IEEE and Scopus databases were considered in the exploration. Results: 50 articles addressing the subject between 2013 and 2022 with growing interest. The core use of this technology is related to networking and health areas, articulating Body sensor networks or data policies management (collecting, routing, transmission, and workload management) with edge computing. Conclusions: Real-time Reliability Monitoring in edge computing is ongoing and still nascent. It lacks standards but has taken importance and interest in the last two years. Most articles focused on Push-based data collection methods for supporting centralized decisionmaking strategies. Additionally, to networking and health, it concentrated and deployed on industrial and environmental monitoring. However, there are multiple opportunities and paths to walk to improve it. E.g., data interoperability, federated and collaborative decision-making models, formalization of the experimental design for measurement process, data sovereignty, organizational memory to capitalize previous knowledge (and experiences), calibration and recalibration strategies for data sources.

Keywords: real-time, system reliability, monitoring, edge computing, systematic mapping study.

1. Introduction. The continuously evolving technological landscape has driven the need for real-time data processing to meet emerging requirements. Consequently, there is a renewed interest in reevaluating real-time concepts and their alignment with advancements and new contexts [1]. These reasons vary, ranging from leveraging up-to-date information about entities to making critical decisions [2-4] and mitigating natural disasters [5]. These advancements have paved the way for various monitoring applications in sectors such as retail [6], industry [7, 8], and natural disaster management [9–11], among others. In applications directly associated with people (e.g., natural disasters and rescue), real-time represents the difference between whether or not keeping a person alive [10]. To achieve this, the Internet of Things (IoT) has played a crucial role by enabling data processing, sensor integration, storage, and communication systems to be deployed as close to the monitored

environment as possible [12, 13]. Simultaneously, Edge Computing (EC) has emerged to enhance resource management and workload processing [14, 15]. This approach has increased computing power, local resource availability, and communication capabilities while enabling effective resource allocation and management within specific events. Various networks, including body sensor networks and road sensor networks, have been developed and deployed as part of a common and heterogeneous infrastructure, each serving different purposes [16].

Real-Time Reliability Monitoring (RTRM), which relies on IoT devices and EC, is based on a crucial assumption – the system is reliable. However, it is important to establish a clear and shared definition of reliability to ensure a common understanding. Reliability refers to the level of achievement at which a system, product, or component meets its functional specifications within specified conditions and time intervals [17]. A reliable system should exhibit maturity, availability, fault tolerance, and recoverability.

Maturity entails meeting the reliability requirements during typical system operation. Availability ensures that the system is operational and accessible whenever it is required. Fault tolerance focuses on maintaining the system's expected functionality even in the presence of hardware or software faults. Recoverability deals with the system's ability to recover affected data and restore the desired state of a component following an interruption or failure event. Therefore, while reliability represents the desired objective, resiliency encompasses the various elements and procedures employed to attain such a status.

In critical areas such as healthcare, real-time analysis relies on the utmost reliability and monitoring. The advent of IoT has enabled the proliferation of sensing applications, leading to increased density and coverage of data collection related to various aspects, such as the impact of particulate matter on people's health. However, it is crucial to acknowledge and address concerns associated with sensor measurements, such as miscalibrations, before making decisions based on such data [18]. The assumptions underlying the measurement process must be precise and treated with the same level of importance as any decision-making process, whether automated or not.

The collection of measures should adhere to certain principles: repeatability (consistent data collection methods), extensibility (ability to accommodate new requirements), clear definition, and alignment with the intended purpose. The measures obtained must be comparable and consistent, reflecting a cohesive and coherent representation of the property being quantified. They should also possess a known degree of accuracy and precision, achieved using calibrated devices (instruments or sensors). When introducing data-driven decision-making [19], decision-makers expect the data to be interesting, consistent, trustworthy, and timely. They seek information, rather than raw data, to inform their decisions across various aspects. This concern becomes even more critical in the context of automated sensor networks, where decision-making processes are automated. Therefore, ensuring system reliability monitoring, along with a robust measurement process, becomes essential.

Given the critical areas (e.g., health) where real-time analysis is applied, reliability and monitoring are mandatory. On the one hand, IoT fostered increasing the density and coverage of sensing applications (e.g., particulate matter and its impact on people's health). On the other hand, sensors suffer from different concerns that need to be considered (e.g., miscalibrations) before deciding based on such data [18]. The underlying assumptions around the measurement process are sharp and must be regarded as in any decision-making process (automated or not). How the measures are collected must be repeatable (i.e., collects data in a compatible way), extensible (e.g., new requirements), defined, and aim guided. The obtained measures must be comparable and consistent (i.e., cohesive and coherent based on the property trying to quantify), with a known degree of accuracy and precision using a calibrated device (instrument or sensor). Always Datadriven decision-making [19] gets introduced, and decision-makers believe that data are interesting, consistent, trust, and timely. Thus, decision-makers expect to use information (and not data) before deciding on any aspect. Now, it extrapolates this concern in automated sensor networks where the decisionmaking is automated. System reliability monitoring (and the underlying measurement process) becomes essential.

The Systematic Mapping Study (SMS) describes a set of guides to perform a study focused on tracing the evidence in a broad domain, such as RTRM on EC. This approach allows for a preliminary exploration of the topic before delving into specific aspects [20 - 22].

This work contributes in several ways: a) an SMS was conducted, providing a detailed identification and characterization of strategies related to RTRM on EC; b) different approaches focusing on reliability monitoring were identified and characterized; c) a comparative analysis was performed to elucidate the relationship between artificial intelligence, reliability, and monitoring on edge devices; d) the reliability monitoring needs and their impact on automated decision-making processes were analyzed within the context of edge computing.

The study is divided into seven sections. Section 2 provides a summary of related articles that share similar objectives. Section 3 outlines the methodological strategy, including the selected data sources, criteria for

article selection, and retention criteria used to compile the final list of articles. Section 4 analyzes the collected documents using the SMS methodology, describing the application of SMS and the analysis of the subjects, along with a scoring model for prioritizing further reading. Section 5 presents a comparative analysis of the main strategies based on the scoring model. Section 6 discusses the results obtained in relation to the research questions. Finally, the study concludes with a section dedicated to presenting the overall findings and conclusions.

2. Related Works. There is a vast number of reviews available on the topics of IoT and EC, particularly related to sensor networks and data gathering for monitoring purposes. However, this section specifically focuses on the most relevant works pertaining to RTRM on EC, which examines system reliability monitoring at an abstract level. The objective is to identify reviews that emphasize reliability as a core concept.

Additionally, the monitoring schema should prioritize the degree to which a system achieves functional specifications under defined conditions within a specific time interval. This means that the system can monitor various entities such as vehicles, trains, roads, outpatients, etc., but the crucial aspect lies in ensuring that the primary monitoring system operates as intended.

In a review by [23], the authors delve into edge computing applications in smart grids and compare their efficiency to traditional systems. They discuss the advantages of employing edge devices in smart meters, which results in improved accuracy, reduced latency, and optimized bandwidth consumption. The article explores various EC-based applications, including the monitoring of transmission lines using Unmanned Aerial Vehicles, video surveillance systems, microgrid systems, and power supply management in charging points. It also discusses RTRM of edge frameworks, the interplay between IoT, EC, and cloud computing, as well as the correlation between productivity and security in smart grids based on EC.

While this review shares similarities in addressing the impact of EC devices on latency, bandwidth, and accuracy, it differs in its primary focus on RTRM as a core concern, regardless of the specific application domain.

In [24], the authors introduce a review of mobile and wearable sensors for data-driven health monitoring systems. They discuss strengths and weaknesses associated with wearable sensors, available alternatives, and monitoring systems. Also, the work addresses data types and issues related to disease diagnosis domains, critical challenges inhibiting sensor-based health monitoring systems, and proposed solutions. As a similarity, it addresses a systematic mapping study on Sensor-based applications and solutions. As a difference, this systematic mapping study focuses on the reliability of EC-based systems and how different works try to achieve defined system requirements. It discusses the articulation between cloud and EC for cooperating on meteorological radar applications and data interoperability [25]. The authors discuss the main problems of weather radar data quality control and different scenarios where edge and cloud computing are complimented for data quality control. The article introduces and contrasts various techniques for data quality control jointly with the challenges and association with edge-cloud cooperation. As a similarity, the review focuses on real-time alternatives to deal with data quality control and the associated implementation (and not the data gathering itself). As a difference, this SMS addresses the system reliability considering different data collection strategies (e.g., push-based, pull-based, hierarchical variations, etc.) jointly with the decision-making process and how to implement actions to keep aligned reliability requirements.

In [26], the authors discuss the impact of information and communication technologies on Digital Twins as a Service (DTaaS). They substantiate the increasing interest in the industrial field due to technologies related to EC, network function virtualization, and 5G. As a similarity, it analyses the latency, data rate, reliability, and scalability requirements in different functionalities related to digital twins (i.e., monitoring, simulation, and operation). However, it limits the level of scalability outlined based on use cases. As a difference, this work discusses the SMS of the RTRM on EC as a central concept, analyzing research volume, types of applications, strategies, and decision-making data collection approaches (e.g., centralized, distributed, etc.).

3. Methodology. SMS proposes a set of directives to specify an action protocol to analyze the shell of a topic. It is beneficial in those areas where a topic is extensive and needs to be reduced based on determined criteria [20 - 22]. This way, after the topic's shell is known and described, it is feasible to explain it before focusing on further detailed analysis.

Figure 1 outlines a Business Process Model and Notation (BPMN) diagram depicting the crucial steps and activities related to the SMS. First, the aim that guided the study was defined jointly with the Research Questions (RQ). An essential aspect was to cover the purpose through RQs before advancing; In other words, the research questions should add different approaches to explore the topic based on the objective. After that, it chooses the databases where the inquiries will be executed. In this sense, every RQ is translated as a part of the search string and suited to the particularities of each chosen database, incorporating the filtering criteria.

This way, the final search strings (suited to each chosen database, including the filtering criteria) were run on each database. Then, the results were processed and reduced through established criteria.



Fig. 1. Crucial Steps of the Systematic Mapping Study, using the BPMN notation

It is worth mentioning that database results were unified in a single list. Then, a quick analysis removed the duplicate records from the list while the selection criteria were defined. The retention criteria focused on the wished properties in articles to sustain the analysis based on coverage and the study's objective. At the same time, those that did not meet the requirements were separated, which helped debug the results and obtain a debugged list.

The next step was to read each article from the debugged list. Records were kept depending upon whether the contents were related to the study objective and the associated points of view (e.g., perhaps an article figured in the list because keywords were present, but the content could mention them as a related example – or potential application – without specific details on the topic). After analyzing every article, a unique list of articles aligned with the objective and proposed questions was produced.

The next step aimed to define dimensions and categories for analyzing those articles from the list. On the one hand, dimensions describe a broad perspective from research questions, while the categories focus on specific content in the dimension. For instance, a dimension could analyze the approaches used for monitoring the system's reliability; however, categories could indicate different typologies for implementing the monitoring, such as "Push-based, Pull-based, distributed, or centralized." Once the dimensions and categories were specified, they helped to classify the articles in the list. Then, articles were summarized, organized, and communicated through the conclusion in alignment with the study aim.

This section is organized into four sections. The first section specifies the study aim and research questions. The second focuses on the search strategy, target databases, and how to suit the search string in each case. The third one discusses the selection process. The last part of this section makes explicit the limitations and potential biases of this study.

3.1. Research Questions. RQ emerged naturally from the defined aim to define the analysis perspectives and describe the broadness of the search strategy. As it was introduced, the objective of this SMS resided in identifying and characterizing the IoT-based RTRM strategies, which have implemented a measurement process to study the effects of the particulate matter on human health.

The first one was *What is the volume of research on Artificial Intelligence (AI) applied to RTRM on edge devices?* (i.e., RQ1). It is focused on exploring the main concentration of AI-related techniques on current reliability concerns and challenges. It is interesting to review and characterize the current trends. Also, it is worth analyzing the reasons behind the topics with the closest attention in contrast to those with the lowest popularity.

The second question was *What kind of AI approaches are used for reliability monitoring on edge devices?* (i.e., RQ2) The inquiry aims to dive into the different approaches used for reliability monitoring. It could be associated with a holistic approach or dive into some reliability subattributes such as maturity, availability, fault-tolerance, or recoverability. Thus, methods could refer to the data processing mode (i.e., online or offline), how metrics are collected (i.e., push or pull), etc.

The third question analyzed the relationship between AI approaches and Data-driven decision-making. It was "*Are the AI approaches related to some real-time decision-making strategy*?" (i.e., RQ3). In this perspective, the underlying assumption implies there is an AI approach guided by data and applied to some situations. Thus, the research question focused on the articulation between the AI model and the Decision-making strategy.

The fourth question advocated for the data collection methods and the articulation with Real-time Reliability Monitoring. This question was *What data collection methods are used for implementing RTRM on eEC?*" (i.e., RQ4). These aspects are worth considering due to the edge challenges such as device heterogeneity, device distribution, data ownership, data sovereignty, etc.

The last research question was *What are the main applications of AI-based RTRM on EC?* (i.e., RQ5). In other words, the idea is to dive into kinds of applications, fields, areas, or subjects where RTRM gets required, and the AI models represent a positive plus. Thus, the characterization of such applications would allow extracting features under interest.

From the research questions, the idea was to identify the concepts involved in each question to delineate the query strings. Thus, once the query strings were identified, they were adapted following the syntaxis of each queried database [21].

Table 1 describes the keywords surged from each research question, defining the alternative terms for known concepts such as Reliability. While the research questions indicate each complementary perspective under analysis based on the aim of the view, keywords allowed for addressing such views converting them into conceptual search strings to locate articles satisfying each view (acting as decision criteria) [22]. Keywords related to AI, RTRM, Strategy, Reliability, and EC emerged from RQ1 to analyze the articulations between the devices, collecting systems, and those AI approaches to monitor them [27 - 29]. Keywords associated with the types and uses emerged from RQ2, intending to analyze the alternative approach for reliability monitoring [30-31]. Keywords about AI, Decision-Making, and Relationship get derived from RQ3 to explore the articulation between knowledge (or previous experiences) and decision [32 - 34]. Keywords related to the data collection use and reliability used from RO4 to analyze the common collection methods and the reliability monitoring [35 - 37]. Keywords associated with the applications surged from RQ5 [38 - 40].

Table 1. Keywords related to Research Questions

RQ	Keywords
	AI (alternatively "Artificial Intelligence," "Machine Learning," "Expert Systems,"
	"Agents," "Case-based Reasoning," "Linked Systems," "Optimization," "Automated
1	Planning and Scheduling," or "Computer Vision"), Real-time, Reliability (alternatively
1	"Maturity," "Availability," "Fault-Tolerance," or "Recoverability"), Monitoring
	(alternatively "Observation" or "Monitor" or "Observability"), Strategy (alternatively,
	"Approach," "Method," "Procedure," "Scheme"), and "Edge Computing".
2	Kind (alternatively, "Class," or "Category," or "Type"), Al ¹ , Approach ¹ , "Used for"
2	(alternatively "Applied to"), <i>Reliability¹</i> , <i>Monitoring¹</i> , and " <i>Edge Computing</i> ".
3	AI ¹ , Approach ¹ , Relationship (alternatively "Association"), Real-time, Decision-making.
4	"Data collection" (alternatively "data gathering"), Strategy ¹ , "used for" ¹ , Real-
4	time, Reliability ¹ , Monitor, "Edge Computing".
5	Applications (alternatively "Use Case"), Al ¹ , Real-time, Reliability ¹ , "Edge Computing".

¹ As defined in previous research questions

Table 2 describes the conceptual search string derived from keywords in Table 1.

Terms	Structure
Common	("Artificial Intelligence" or "AI" or "Machine Learning" or "Expert Systems" or "Agents" or "Case-based Reasoning" or "Linked Systems" or "Optimization" or "Automated Planning and Scheduling" or "Computer Vision") and "Real-time" and ("Reliability" or "Maturity" or "Availability" or "Fault-Tolerance" or "Recoverability") and ("Monitoring" or "Monitor" or "Observation" or Observability) and ("Strategy" or "Approach" or "Method" or "Procedure" or "Schema") and ("Edge Computing" or "Edge" or "IoT" or "Internet-of-Thing")
Alternative	("Kind" or "Class" or "Category" or "Type") or ("Used for" or "Applied to") or ("Relationship" or "Association") or ("Decision-making") or ("Data collection" or "data gathering") or ("Applications" or "Use Case")

Table 2. Conceptual Search String

Table 2 discriminates between common and alternative restrictions for logical expression. On the one hand, the common terms must be simultaneously present in the abstract, title, or keywords. On the other hand, the alternative terms represent different concepts eventually associated. However, at least one of them must be present in the abstract, title, or keywords jointly with the familiar words.

Thus, an article was retained in the initial result if it contained all the common terms plus at least one alternative term. It is essential to mention that Table 2 describes the conceptual search string, which was fitted to each database's syntaxis before applying it.

3.2. Search Strategy. The conceptual search string was fitted to each database where the query would be performed. For this study, the digital libraries of IEEE and Scopus were considered. The idea was to provide broad coverage around the published articles related to the subject.

Tables 3 and 4 describe the adaptation of the conceptual search string to the Scopus and IEEE database syntaxes. As it is possible to appreciate, even when the meaning is similar, some syntactic aspects were fitted before running the query.

Table 3. Search String fitted to Scopus Database

Search String
TITLE-ABS-KEY ("Artificial Intelligence" OR "AI" OR "Machine Learning" OR "Expert
Systems" OR "Agents" OR "Case-based Reasoning" OR "Linked Systems" OR "Optimization"
OR "Automated Planning and Scheduling" OR "Computer Vision") AND TITLE-ABS-KEY
("Real-time") AND TITLE-ABS-KEY ("Reliability" OR "Maturity" OR "Availability" OR
"Fault-Tolerance" OR "Recoverability") AND TITLE-ABS-KEY ("Monitoring" OR
"Monitor" OR "Observation" OR "Observability") AND TITLE-ABS-KEY ("Strategy" OR
"Approach" OR "Method" OR "Procedure" OR "Schema") AND TITLE-ABS-KEY ("Edge
Computing" OR "Edge" OR "IoT" OR "Internet of Thing") AND TITLE-ABS-KEY ("Kind"
OR "Class" OR "Category" OR "Type" OR "Used for" OR "Applied to" OR "Relationship"
OR "Association" OR "Decision-making" OR "Data collection" OR "data gathering" OR
"Applications" OR "Use Case")

Table 4. Search String Fitted To IEEE Database

Search	String
Scaren	Stime

("All Metadata": "Machine Learning" OR "All Metadata": "Artificial Intelligence" OR "All Metadata": "AI" OR "All Metadata": "Expert Systems" OR "All Metadata": "Agents" OR "All Metadata": "Case-based reasoning" OR "All Metadata": "Linked Systems" OR "All Metadata": "Optimization" OR "All Metadata": "Automated Planning and Scheduling" OR "All Metadata": "Computer Vision") AND ("All Metadata": Reliability OR "All Metadata": Maturity OR "All Metadata": Availability OR "All Metadata": "Fault-Tolerance" OR "All Metadata": Recoverability) AND ("All Metadata": Monitoring OR "All Metadata": Monitor OR "All Metadata": Observation OR "All Metadata": Observability) AND ("All Metadata": Real-time) AND ("All Metadata": "Edge Computing" OR "All Metadata": "Edge" OR "All Metadata": "IoT" OR "All Metadata": "Internet of Thing") AND ("All Metadata": Strategy OR "All Metadata": Approach OR "All Metadata": Method OR "All Metadata": Procedure OR "All Metadata": Schema) AND ("All Metadata": Kind OR "All Metadata": Class OR "All Metadata": Category OR "All Metadata": Type OR "All Metadata": "Used for" OR "All Metadata": "Applied to" OR "All Metadata": Relationship OR "All Metadata": Association OR "All Metadata": "Decision-making" OR "All Metadata": "Data collection" OR "All Metadata": "Data gathering" OR "All Metadata": "Applications" OR "All Metadata": "Use case")

3.3. The Selection Process. This SMS focused on consolidated research written in English and published in journals. It did not consider those articles related to conferences (except extended versions), abstracts, posters, general surveys, reviews, or similar. Thus, it assumes the consolidated research is published in journals and written in English, pursuing international visibility. In this sense, the idea focused on getting an updated perspective about consolidated research strongly associated with Edge-aware systems for RTRM. For sure, this decision implies a bias around the review.

Thus, the initial result list was filtered, retaining articles written in English and coming from journals. Also, surveys and reviews were kept out, fostering access to the source.

Once results from each database were obtained and filtered, a unique list was generated from the fusion of the previous ones. Over this new list merging results from each database, only one copy of duplicate records was kept.

After that, each article's abstract was read in the list to confirm whether it was retained or not according to the search string. That is to say, the searched terms (i.e., keywords) could be present in the abstract just as a mention but not as a core subject. Just articles addressing as a core subject those aspects indicated in the keywords introduced by research questions (Table 1) were retained. In this way, a debugged list for its analysis was reached. This aspect incorporated subjectivity derived from the author's criteria in this work, which was understood as a potential bias.

3.4. Biases and Limitations. The primary underlying assumption throughout this work was that consolidated results from research lines were published in journals. For this reason, no conference papers and other kinds of publishing were considered. Also, previous reviews were supposed to

only partially analyze the core aspect of real-time system reliability, avoiding intermediaries.

English is assumed to be the common language to internationalize the research results.

The limitation of the systematic review is associated with biases. The inclusion/exclusion criteria and the selection of databases compound a set of elements that could be naturally subjective due to the authors' perception. This work was limited to articles published in journals and available on IEEE and Scopus databases. Even when the work tried to get a broad scope, it is not a warranty that the universe of articles is considered.

4. Description of the Approaches. This section is broken down into three sections to describe the implementation of SMS, jointly with the analysis of topics and the proposed order to introduce each obtained work from the queries.

4.1. Implementing the SMS. The queries ran in their respective databases on December 14, 2022. After the initial results, the indicated filters were applied (i.e., limiting to journal articles written in English, etc.), obtaining a partial list of a) *IEEE*: 22 records and b) *Scopus*: 42 records. In this way, 64 records initially satisfied the search requirements.

Each query's result was stored in an Excel file organized by the origin database, and each item was individually inspected to verify that it was associated with a journal article. If a record was not associated with a research article from a journal, it was removed from the list.

Table 5 shows the reasons and quantities for each record's retention jointly with the removal. On the one hand, those records referred to RTRM based on EC were retained. On the other hand, documents associated with the following types were removed 1) *Duplicated*: The same article was available in the IEEE and Scopus results. Only one copy was retained; 2) *Not related content*: The article specifically did not refer to RTRM based on EC. Keywords were only mentioned in the narrative; 3) *Proceedings*: It was not an article but a conference proceeding; 4) *Review*: The article was not a research article. It contained a survey or review of the subject but not how to monitor the system reliability through EC in real-time.

	Scopus	IEEE
Retained	34	16
Duplicated	0	6
Not Related	8	0
Proceedings	0	0
Review	0	0
Total	42	22

Table 5. Results

Thus, 58 records were retained from the different databases, removing eight duplicates and obtaining 50 unique articles (See the first line under the caption in Table 5).

4.2. Dimensions and Characteristics. From the research questions, it was possible to define the dimensions associated with each one jointly with the categories to classify each record from the 50. Table 6 describes the dimensions and categories established by each research question to address the synthesis process indicated in Figure 1.

Research Question	Dimension	Categories
RQ1	Research	Artificial Intelligence, Real-time, Reliability, Monitoring, Edge Devices
RQ2	Approach	Push-based, Pull-based, Distributed, Centralized
RQ3	Decision- Making	Centralized, Federated, Collaborative
RQ4	Methods	Direct Push, Direct Pull, Hierarchical Push, Hierarchical Pull, Query-based local
RQ5	Applications	Area, Subject

Table 6. Dimensions and Categories

About RQ1, the approaches were 1) "AI" when models are applied to monitor, analyze, or forecast some properties associated with the system reliability, 2) "Real-time" when data are processed and analyzed when they arrive, 3) "Reliability" when the monitoring strategy monitors the system reliability itself in addition to the application aim, 4) "Monitoring" indicates a formalized strategy to observe (continuously and actively) the system's reliability, 5) "Edge Computing" when edge devices are articulated with the monitoring strategy focused on the system reliability. RO2 refers to the approach used for monitoring the system's reliability. It could be "Pushbased" when the edge devices push data to a reliability monitoring component; "Pull-based" when a reliability monitoring component scratches data from the edge devices to ensure the system's reliability; "Distributed" when the reliability monitoring is associated with some swarm intelligence where data are kept on the origin; "Centralized": It indicates an approach where system reliability monitoring is associated with a central component, and edge devices play the role of obedience. RQ3 was associated with the decision-making strategy, indicating "Centralized" for those strategies where a significant piece makes decisions based on collected data, "Federated" when the decision-making strategy implies a set of autonomous and distributed features which decide based on some

consensus and feedback mechanism between them, and "Collaborative" is when two or more components exchange data and knowledge optionally before choosing between them. In this approach, collaboration is optional. A system component could act unilaterally or with a few other devices. RQ4 was focused on the data collection strategy and decision-making process. "Direct Push" indicates data are pushed to a central reliability monitoring component without intermediaries or data aggregation. "Direct Pull" is associated with a centralized reliability monitoring component scratching data directly from the data sources. "Hierarchical Push" is when data walks through a data processing hierarchy from the data source until reaching the wished reliability control level. "Hierarchical Pull" focuses on data traveling in a data processing hierarchy, but a certain reliability control level pulls data from the source in stages. "Query-based local" indicates that data resides locally (never moved), and the data source only answers queries using the local data. Finally, RQ5 was related to the applications. Thus, "Area" represents the field in which the system reliability monitoring is applied, while "Subject" refers to the specific application.

Using the ISSN (International Standard Serial Number) associated with the journal of each selected article, the quarter to which the journal belongs was obtained from Scimago Journal and Country Rank (SJCR) [41] and incorporated for the analysis. SJCR is a publicly available portal that includes the journals and country scientific indicators created using information from the Scopus database (Elsevier B.V.). The platform uses the Scimago Journal Rank indicator [42] to show the visibility of journals in the Scopus database from 1996. The indicator considers the journal prestige, the number of associated documents in the database, and the citation prestige given by "importance" and "closeness" according to the received citations. Thus, journals could fall into four categories a) Quarter 1 (Q1) indicates the Top-25% percent most visible journals for an area or discipline, 2) Quarter 2 (O2) represents journals located in the region of (25; 50] %, 3) Quarter 3 (Q3) indicates those journals integrating the zone of (50; 75] %, and 4) Quarter 4 (Q4) composed of journals falling into the interval (75; 100] %. Eventually, it is possible that a journal is not registered or present in the Scopus database (e.g., because it is a new journal); in such a case, missing space is indicated.

Figure 2 describes the distribution by year of the retained articles from the queries, indicating the quarter to which the associated journal belongs. Since 2013 this field has taken sustained growth. As mentioned, the basic Scimago's journal categories go from Q1 to Q4. However, some journals contain a mix of types (e.g., Q1/Q2), indicating that some journal subjects are assessed and fall under a specific category (e.g., Q1). Still,

other ones fall under a different category (e.g., Q2). Some journals do not have a Scimago category because they are new. The previous figure indicates this situation under the "Missing" category.



Through the individual reading of each article, they were classified according to the categories specified for each dimension. Regarding the research dimension, 74% of analyzed works incorporated some AI model, 94% focused on the real-time aspects, 90% analyzed specific reliability properties, 70% described a formalized monitoring strategy, and 98% explained the articulation between edge devices to IoT and Cloud computing.

The dimension related to the reliability monitoring approach shows that 75.86% of works follow a Push-based strategy. It is logical because there is no control over exogen factors affecting the deployment field (or decisions from third parties) in a distributed and heterogeneous environment. 12.07% of works incorporate a central component for system reliability monitoring. 10.34% of analyzed documents describe a Pull-based associated with the system reliability monitoring component. Lastly, 1.72% of works outline a distributed strategy for dealing with system reliability monitoring.

About the Decision-making dimension, 71.43% of works are associated with a centralized schema to make decisions on system reliability, 17.86% describe a distributed approach, and 10.71% introduced a federated strategy.

From the point of view of data collection methods, 44.23% address the data collection using direct-push strategies, and 42.31% describe a hierarchical-push approach (e.g., using gateways or variations of aggregation

schemas). It represents 86.54% of works using push strategies to provide data for system reliability monitoring. 7.69% of results deal with pull-based collection schemas, and 3.85% use a hierarchical-pull approach (i.e., 11.54% of works address the data collection using pull-based methods). 1.92% of results incorporate local data management articulated with the system reliability monitoring component through a Query-based schema.

Figure 3 describes the proportion of application areas based on the general area concept characterized by the fifth dimension.



Fig. 3. Main Application Areas

5. A comparative perspective of the main approaches. Each of the obtained articles was synthetically described and analyzed from the dimensions introduced in Table 5. For better organizing, they will be presented according to the generated groups in the previous section and following the order criteria.

5.1. Articles between 2013 and 2019. Table 7 describes articles between 2013 and 2019, ordered by year, quarter, and the number of citations, indicating the decision-making approach.

Article	Decision-	Area	Scimago	Citation
	making	I ti ca	Sennago	Citation
IOT system environmental monitoring using IPSO weight factor estimation [43]	2013	Centralized	System environmental monitoring	Q3
Self-regulating supply–demand systems [44]	2017	Collaborative	Energy supply- demand Matching	Q1
Real-Time Signal Quality-Aware ECG Telemetry System for IoT- Based Health Care Monitoring [45]	2017	Centralized	ECG Telemetry System	Q1
PsCPS: A distributed platform for cloud and fog integrated smart cyber- physical systems [46]	2018	Centralized	Cyber-physical Systems	Q1
Contextaide: End-to-End Architecture for Mobile Crowd- sensing Applications [47]	2019	Centralized & Collaborative	Perpetrator Tracking	Q1
Fault-Tolerant mHealth Framework in the Context of IoT-Based Real-Time Wearable Health Data Sensors [48]	2019	Federated & Collaborative	mHealth. Reliability	Q1
QoS-Adaptive Approximate Real- Time Computation for Mobility- Aware IoT Lifetime Optimization [49]	2019	Centralized	Task Schedule based on QoS requirements.	Q1/Q2
Deep Learning for Hybrid 5G Services in Mobile Edge Computing [50]	2019	Centralized	Network. Energy consumption optimization	Q1

Table 7. Prioritizing for Articles between 2013 and 2019

Study [43] describes an improvement in particle swarm optimization (PSO). The authors claim that Improved PSO increased the measurement preciseness via weight factors estimated via experimental simulations. It allows the integration of weight factors, information source fusion reliability, information redundancy, and hierarchical structure integration in uncertain fusion cases.

In [44], the authors describe a generic decentralized self-regulatory framework shaped around standardized control system concepts and IoT. It involves a decentralized combinatorial optimization mechanism that matches supply-demand under different regulatory scenarios. Also, they describe an evaluation methodology jointly with the proposal that allows the systematic assessment of constraints.

In [45], it is described the signal quality-aware IoT-enabled electrocardiogram (ECG) telemetry system for continuous cardiac health monitoring applications. It comprises modules such as ECG signal sensing, automated Signal Quality Assessment (SQA), and signal-quality aware (SQAw) ECG analysis and transmission. It aims to design and develop a lightweight ECG SQA method for Real-time classifying based on ECG signals.

It proposes a distributed platform for cloud and fog-integrated smart Cyber-Physical Systems (sCPS), named PsCPS [46]. It can provide services to address integration challenges among multiple clouds, multiple fog nodes, and sCPS subsystems. It comprises agents providing services for cloud and fog-integrated sCPS in the participating nodes. Agents can be developed, implemented, controlled, and managed in different ways (i.e., a set of single agents, multi-agent systems, or hierarchical multi-agent systems).

In [47], the authors introduce a design approach for tracking using Mobile-Crowd Sensing (MCS) for complex processing in real-time named ContextAiDe. It comprises 1) An API for detailing contexts jointly with the MCS applications, 2) A middleware for enabling context-optimized and fault-tolerant distributed executions, and 3) An optimization engine for providing the action paths. The authors claim that the strategy is contextoptimized recruitment for the execution of computation- and communication-heavy MCS applications in an edge environment. It analyzes two contexts: a) Exact: which has a set of constraints that must be satisfied, and b) Preferred: which has constraints that may be satisfied to a certain level. Thus, ContextAiDe can optimize the operational overheads to enable real-time operation by adjusting the preferred contexts.

In [48], the authors describe a framework to deal with different concerns associated with mHealth applications. They explain a risk local triage algorithm (Risk-Level Localization Triage – RLLT). It can exclude the control process of patient triage and warn about failures related to wearable sensors. RLLT carries out the initial step towards detecting a patient's emergency case and then identifying the healthcare service package of the risk level. Thus, decision-makers are assisted with hospital selection based on the risk-level package, considering the time to arrive at every hospital related to the patient. Also, a mobile app can connect with the hospital's servers to verify the availability of healthcare services given a risk-level package to guide the feasible options.

In [49], the authors address the mobility-aware network lifetime maximization for battery-powered IoT applications. They focus on estimating real-time computation subject to quality-of-service constraints. They discuss a two-fold strategy. An offline step where a task schedule based on mixed-integer linear programming tries to optimize the network lifetime. An online step deals with a heuristic approach for adapting the task execution based on QoS requirements.

In [50], the authors introduce a strategy to minimize normalized energy consumption. In this context, energy consumption refers to the consumption per bit. The optimization is pursued through optimizing user association, resource allocation, and offloading probabilities subject to the quality-of-service requirements. On the one hand, the mobility management entity manages the user association. On the other hand, every access point determines the resource allocation and offloading likelihoods. A deep neural network is trained offline in a central server based on a digital twin approach representing the current state of the network. Thus, the Mobility management entity uses a DNN model to obtain the user association scheme in real time. Next, an optimization algorithm analyzes the best resource allocation and offloading likelihoods at each access point.

5.2. Articles in 2020. Table 8 describes articles in 2020 ordered by quarter and the number of citations, indicating the decision-making approach.

Article	Decision- making	Area	Scimago	Citation
Joint Optimization of Offloading Utility and Privacy for Edge Computing Enabled IoT [51]	Centralized & Collaborative	Network. Resource Optimization	Q1	102
Towards collaborative intelligent IoT eHealth: From device to fog, and cloud [52]	Centralized & Collaborative	ECG-based arrhythmia detection	Q2/Q3	47
A Joint Deep Learning and Internet of Medical Things Driven Framework for Elderly Patients [53]	Centralized	Elderly Patients Monitoring	Q1	46
AI-Enabled Reliable Channel Modeling Architecture for Fog Computing Vehicular Networks [54]	Centralized	Vehicular Monitoring	Q1	26
Quantum computing-inspired network optimization for IoT applications [55]	Centralized	Sensor Networks > Improve the RT Data accuracy	Q1	23
Research on environmental monitoring trend analysis based on internet of things visualization technology [56]	Centralized	Environmental Monitoring	Q4	1

Table 8. Prioritizing for Articles in 2020

In [51], it gets described as a two-phase offloading optimization strategy. It aims for joint optimization of offloading utility and privacy in the Edge. The Resource utilization maximization is reached through a Utility-Aware Task Offloading (UTO) method. On the other hand, a balance between privacy preservation and execution performance is achieved using a mutual optimization method.

In [52], it describes a holistic AI-driven IoT eHealth architecture based on the concept of the Collaborative Machine Learning approach. Intelligence is distributed across different layers: Devices, Edge/Fog, and Cloud. The proposed approach enables healthcare professionals to continuously monitor the health-related data of subjects anywhere at any time, providing real-time actionable insights for improving decision-making. The proposal is applied to the ECG-based arrhythmia detection case study.

In [53], the authors describe a sustainable, reliable, and optimized strategy for cardiac activity monitoring based on wearable sensors and image processing in elderly patients. A self-adaptive power control-based Enhanced Efficient-Aware (EEA) approach improves energy consumption and battery lifetime. In this context, it introduces a framework and architecture for cardiac image processing based on deep learning and the Internet of Medical Things. Sustainability and reliability are described around a model for optimizing battery usage and network optimization.

In [54], the authors propose an AI-based, Reliable Interference-free Mobility Management Algorithm (RIMMA) for fog computing intravehicular networks. Contributions are organized around four main axes. The first one they claim is that RIMMA, jointly with fog computing, improves computation, communication, cooperation, and storage space. The second one, it gets proposed a reliable and delay-tolerant wireless channel model with better QoS for passengers. The third one is a reliable and efficient multi-layer fog-driven inter-vehicular framework. The fourth one works on the QoS optimization based on mobility, reliability, and packet loss ratio.

In [55], a Quantum Computing-inspired Optimization (IoT-QCiO) technique is introduced. It focuses on maximizing Data Accuracy (DA) in a real-time environment of IoT applications. The proposed model adds quantum formalization of sensor-specific parameters to quantify IoT devices in terms of Sensors in the Vicinity (SIV) and Optimal Sensor Space (OSS). The algorithm optimization is based on data cost, accuracy, and temporal efficiency.

In [56], it introduces an environmental monitoring trend analysis algorithm based on the Internet of Things visualization technology. The proposed algorithm first divides the environmental monitoring area into several different clusters. Different types of sensors provide environmental data that are associated with a group. A mobile agent node is responsible for a monitoring area. It aims to build a two-dimensional positioning table based on the network energy consumption. Thus, it applies the Rosen gradient projection method for choosing the optimal path. With such information, it builds the mobile plan for data collection and forwarding to the control center. Finally, data are centrally analyzed by the control center.

5.3. Articles in 2021. Table 9 describes articles in 2021 ordered by quarter and the number of citations, indicating the decision-making approach and application area. Works with at least one mention are incorporated into the table, and the other ones are outlined in this section after those related to the table are presented.

Table 9. Priori	Table 9. Prioritizing for Articles in 2021						
Article	Decision- making	Area	Scimago	Citation			
Digital electronics in fibers enable fabric-based machine-learning inference [57]	Centralized	Digital Electronics in fibers	Q1	27			
A Smart, Efficient, and Reliable Parking Surveillance System with Edge Artificial Intelligence on IoT Devices [58]	Centralized	Parking Surveillance	Q1	26			
Enabling Secure Authentication in Industrial IoT with Transfer Learning Empowered Blockchain [59]	Federated & Collaborative	Secure Authentication	Q1	20			
Development of a speed invariant deep learning model with application to condition monitoring of rotating machinery [60]	Centralized	Machinery Clever Monitoring.	Q1	15			
IHSF: An Intelligent Solution for Improved Performance of Reliable and Time-Sensitive Flows in Hybrid SDN- Based FC IoT Systems [61]	Federated	Congestion Control	Q1	11			
An IoT-based deep learning approach to analyze indoor thermal comfort of disabled people [62]	Centralized	Indoor Monitoring	Q1	10			
Fog-centric IoT based smart healthcare support service for monitoring and controlling an epidemic of Swine Flu virus [63]	Centralized	Monitoring of Swine Flu Virus	Q2	8			
Intelligent system of game-theory- based decision making in smart sports industry [64]	Centralized	Athlete Performance Monitoring	Q1	5			
Learning Spatiotemporal Latent Factors of Traffic via Regularized Tensor Factorization: Imputing Missing Values and Forecasting [65]	Centralized	Road Status Forecast	Q1	5			
Hybrid Auto-Scaled Service-Cloud-Based Predictive Workload Modeling and Analysis for Smart Campus System [66]	Centralized	Scalability. Workload Management	Q1	5			
Real-time energy consumption detection simulation of network node in internet of things based on artificial intelligence [67]	Centralized	Network. Resource Monitoring	Q1	4			
Monitoring Cyber SentiHate Social Behavior During COVID-19 Pandemic in North America [68]	Centralized	Online Social Network Monitoring	Q1	3			
Random Forest for Data Aggregation to Monitor and Predict COVID-19 Using Edge Networks [69]	Centralized	COVID-19 Monitor and Prediction	Missing	3			
Scheduling Observers over a Shared Channel with Hard Delivery Deadlines [70]	Centralized	Sensor Network Monitoring & Optimizing	Q1	2			

Table 9	Prioritizing	for Articles	in 2021
1 4010 7.	THOMALING	101 / fitteres	III 2021

		Comm	iaiion oj	Tuble)
Article	Decision- making	Area	Scimago	Citation
Energy-Aware Distributed Edge ML for mHealth Applications with Strict Latency Requirements [71]	Federated & Collaborative	Epileptic seizures prediction	Q1	2
Information Security Monitoring and Management Method Based on Big Data in the Internet of Things Environment [72]	Centralized	Environmental Monitoring	Q1	2
Task Allocation Mechanism for Cable Real-Time Online Monitoring Business Based on Edge Computing [73]	Centralized	Cable Real-time Online Monitoring Business	Q1	1

Continuation of Table 9

In [57], the authors discuss the fabrication approach of digital fibers and its impact on Data Sensor-based decision-making using machine learning. Measures are collected from devices in the thread and stored locally in the same yarn. It allows reaching intra-fiber communications between devices via digital signals. A pre-trained CNN model gets embedded in the fiber and articulated to local data for decision-making.

In [58], the authors discuss the feasibility of using edge computing for intelligent parking occupancy detection using the real-time video feed. It incorporates AI at the edge by implementing an enhanced Single-Shot multi-box Detector (SSD).

The article introduces an authentication strategy supported by Transfer Learning and Blockchain (ATLB) [59]. On the one hand, blockchain technology is applied to privacy preservation and avoids central control in industrial applications. On the other hand, it builds trustworthy blockchains for privacy preservation in industrial applications. In terms of privacy preservation, it introduces different blockchains (e.g., inner and outer) for user authentication strategies to deal with collusion and Sybil attacks. A user-per-region credit approach allows for improving authentication accuracy. It uses transfer learning to minimize the training times between regions.

In [60], the authors describe dealing with model invariance to changing speed via a deep learning method. It can detect a mechanical imbalance (i.e., targeted fault) under varying speed settings. They study speed invariance by processing experimental data obtained from a motor test bed. Also, time-series and time-frequency data are applied to long short-term memory and convolutional neural network, respectively, to evaluate their performance.

In [61], the authors introduce an approach named an Intelligent Solution for Improved Performance of Reliable and Time-sensitive Flows (IPRTF) in Hybrid SDN-based Fog (IHSF) computing IoT systems. The proposal is composed of three solutions. The first one is related to an algorithm to deploy Software-defined Network (SDN) switches between legacy switches to improve network observability. The second one is associated with a {K}-nearest neighbor regression algorithm to Real-time prediction of the legacy links reliability at the SDN controller based on historical data. Thus, the SDN controller can make timely decisions, improving system performance. The third focuses on a Reliable and Time-Sensitive Deep Deterministic Policy Gradient (RT-DDPG) algorithm. It optimally computes forwarding paths in hybrid SDN-F for time-critical traffic flows generated by IoT applications.

In [62], a new learning model is proposed using a deep neural network. The outlined model can predict the indoor thermal comfort of people with different abilities in real time to facilitate remote monitoring. The data collection approach can focus on targeted data before moving them to cloud servers for further analysis.

In [63], the authors explain a fog-centric IoT-based intelligent healthcare support service to monitor and control the Swine Flu virus epidemic. They introduce a framework articulated with fog computing for delay-sensitive applications. Also, they describe the feasibility of using a hybrid classifier to classify the swine flu patient at an early stage and generate alerts to the health officials and patients' guardians.

In [64], an IoT-inspired framework gets proposed for real-time athlete performance analysis. Data from IoT devices represent the base for quantifying the athlete's performance in terms of probability parameters of Probabilistic Measure of Performance (PMP) and Level of Performance Measure (LoPM). The authors introduce a two-player game-theory-based mathematical framework for efficient decision modeling by monitoring officials.

In [65], the authors address the challenge of missing data or noisy information in the context of real-time monitoring. After addressing the previous concerns, the underlying idea is to forecast a city's road status (e.g., congestion). A directed graph models the road network; the nodes are intersections, while the edges represent road segments. The authors assume a set of sensors deployed in the city aligned with the road network. They propose a Temporal Regularized Tensor Factorization (TRTF) framework to account for the spatial structure and temporal dependencies. Also, they describe a data-driven graph-based autoregressive model where weights are learned from data to account for positive and negative correlations.

In [66], it gets introduced a bursts-aware auto-scaling strategy for detecting bursts on dynamic workloads based on resource estimation, decision-making scaling, and workload prediction. The authors pursue

keeping the Quality of Service (QoS) through a hybrid auto-scaled service cloud model dealing with horizontal and vertical scaling concerns. It is addressed through an ensemble algorithm for estimating defined workloads. It allows anticipating the resource management associated with workloads and load balancing for horizontal auto-scaling.

In [67], an artificial intelligence-based Internet of Things control method is proposed. It is aware of the information transmission topology structure to analyze the total energy consumption of the node during transmission. Thus, it estimates the load in each communication cycle to study the energy consumption control of multiple and single transmission path nodes. It fosters the improvement of the resource schedule and energy consumption in mobile network nodes.

In [68], the authors describe a framework for online social network real-time monitoring. It allows for data acquisition, processing, interpreting, and decision-making on the fly when data arrives. Bidirectional Encoder Representations from Transformers (BERT) are applied for natural language processing. BERT-based classifiers discriminate hate and sentiments considering iconic emotions (e.g., emojis). Supervised and unsupervised models are complementarily used for online analysis the social behavior.

In [69], the authors describe an e-healthcare framework based on edge computing. It aims to monitor online health data to predict the risk level of COVID-19 patients. A gateway is incorporated to synthesize data based on the original ones, transmitting those meaningful based on the statistical mean. It allows for minimizing data transmissions from every patient (saving energy), optimizing bandwidth use, and avoiding unnecessary redundancy. Geo-distributed edge servers are used to predict the risk level of each patient with Random Forest (RF) using the synthesized data.

In [70], the authors introduce a framework to formulate the Observer Selection Problem (OSP) through which the controller schedules a sequence of observations that maximize its knowledge about the system's current state. The transmission among observers, controllers, and actuators gets carried out through ultra-low-latency wireless communications. The proposed algorithm systematically prunes the search space to improve the knowledge based on the current situation. The authors claim that its work differs from others because of the extent of the controller's knowledge about the state of the system it controls. Alternative approaches work on real-time communications in that communication reliability is monitored by packet loss or error rate.

In [71], the authors describe an approach for gathering latency requirements of mHealth applications based on user equipment and edge server computing under energy constraints. On the one hand, the authors

address an optimization problem to locate the feature extraction and classification process cleverly between the user equipment and edge devices. It is a Power-aware optimization approach where the inferential mechanism is as close to the user as possible.

Paper [72] describes a framework for real-time environmental monitoring based on IoT. It addresses the basic principles of constructing the evaluation index system while introducing and outlining a reference architecture.

In [73], the authors describe a task allocation mechanism for cable real-time online monitoring business based on edge computing. They define a task allocation model based on the linear distribution characteristics of the cable, the edge node's statuses, the task processing overhead, and the scheduling strategy of delay-sensitive tasks. Next, they introduce a task allocation strategy based on improved discrete particle swarm optimization. The authors claim that the system focuses on the job queuing problem in edge nodes and the optimized task allocation problem among edge nodes.

In [74], it gets presented a strategy for generating a Device-Specific Identifier (i.e., IoT-ID). It captures the device's characteristics and can be used for device identification. The underlying assumption is that IoT-ID is based on Physically Unclonable Functions (PUFs). Thus, it is possible to exploit variations in the manufacturing process to derive a unique fingerprint for integrated circuits. The authors claim that the strategy is non-invasive and can be invoked using simple software APIs running on components of COTS (Commercially Off the Shelf). Among the mentioned properties, they highlight the following: constructability, real-time, uniqueness, and reproducibility.

5.4. Articles in 2022 and the beginning of 2023. Table 10 describes articles in 2022 and the beginning of 2023 that have received at least one citation. This section will initially describe those articles. Next, it is outlined the remaining works (without received citations).

In [75], the authors refer to the impact of the Point of Presence (PoP) in healthcare and how it gains importance as wearable devices and mobile apps are entrusted with RTRM and diagnosis of patients. Novelty relies on the utility value of sensors data improvement through the Laplacian mechanism of preserved Personally Identifiable Information (PII) response to each query from the Edge Open and Distance Learning (ODL).

In [76], the authors address a framework to optimize energy efficiency, battery lifetime, and reliability for intelligent and connected healthcare. It explains the Adaptive Transmission Data Rate (ATDR) mechanism that works on the average constant energy consumption by varying the active time of the sensor node. It allows for optimizing the energy over the dynamic wireless channel. It explains a Self-Adaptive Routing Algorithm (SARA) to adopt a dynamic source routing mechanism with an energy-efficient and shortest path.

Article	Decision- making	Area	Scimago	Citation
Emphasizing privacy and security of edge intelligence with machine learning for healthcare [75]	Federated	Patient Monitoring	Q2	13
AI-driven adaptive reliable and sustainable approach for internet of things enabled healthcare system [76]	Centralized	Body Sensor Network	Q2/Q3	8
Smart IoT and Machine Learning- based Framework for Water Quality Assessment and Device Component Monitoring [77]	Centralized	Water Quality Assessment	Q1	5
An innovative edge-based Internet of Energy solution for promoting energy saving in buildings [78]	Centralized	Domestic Energy Saving	Q1	5
Data fusion-based machine learning architecture for intrusion detection [79]	Centralized	Intrusion Detection	Q2	5
Real-Time Fault Detection and Condition Monitoring for Industrial Autonomous Transfer Vehicles Utilizing Edge Artificial Intelligence [80]	Federated	Industrial Autonomous Transfer Vehicles	Q1/Q2	3
Clinical Care of Hyperthyroidism Using Wearable Medical Devices in a Medical IoT Scenario [81]	Centralized	Clinical Care of Hyperthyroidism	Q2	2
MABASR - A Robust Wireless Interface Selection Policy for Heterogeneous Vehicular Networks [82]	Centralized	Selection Policy for Heterogeneous Vehicular Networks	Q1	1
Online Partial Offloading and Task Scheduling in SDN-Fog Networks with Deep Recurrent Reinforcement Learning [83]	Collaborative	Workload Orchestration	Q1	1
Hybrid multi-objective-optimization algorithm for energy efficient priority- based QoS routing in IoT networks [84]	Collaborative	Congestion Control	Q2	1
A Novel CNN-TLSTM Approach for Dengue Disease Identification and Prevention using IoT-Fog Cloud Architecture [85]	Centralized	Dengue Disease Monitoring	Q2/Q3	1

Table 10. Prioritizing	for Articles	in 2022 – 2023
------------------------	--------------	----------------

In [77], an IoT-based real-time framework is outlined to perform water quality management, monitoring, and alerts for taking actions based

on contamination and toxic parameter levels, and device and application performance as the first part of the proposed work. Machine learning models analyze water quality trends and device monitoring and management architecture. It is supported by a data push strategy associated with dashboards and rules working on a cloud platform.

In [78], it gets introduced the M2SP-EdgeIoE system focused on domestic energy saving and monitoring. It contemplates data collection, Databased non-intrusive load monitoring, anomaly detection (using Autoencoders), and a data-guided recommender system for improving energy consumption.

In [79], the authors discuss the methodology of Real-Time Sequential Deep Extreme Learning Machine (RTS-DELM) implemented in wireless IoT enabled sensor networks for the detection of any intrusion activity. They claim to reach an approach that not only makes the casting of parallel data fusion networks but also renders their computations more effective.

In [80], it gets explained as a generic real-time fault diagnosis and condition monitoring system. It uses edge AI complementarily to the monitoring and proposes a data distributor open-source middleware platform called FIWARE. The application field is related to those autonomous transfer vehicle (ATV) types of equipment targeting a smart factory use case. Anomaly detection in ATVs is analyzed through a deep learning-based fault diagnosis method performed in the Edge AI unit. At the same time, results are sent to the storage through a data pipeline.

In [81], the authors describe a monitoring strategy monitor for patients with hyperthyroidism based on wearable medical devices. An architecture based on gateways articulates different levels of devices that communicate fused data streams (enriched by a metadata-based data model). It is supported by MQTT and RTMP protocols for messaging and video transmissions.

In [82], the authors describe a Multi-Armed Bandit Adaptive Similarity-based Regressor (MABASR) for interface selection. The interface selection policy is analyzed and learned through the relationship between channel quality parameters and uplink data rate.

Study [83] proposes a computational offloading and CPU resource scheduling method for minimizing energy consumption. It considers offloadable and non-off-loadable tasks to characterize and optimize the allocation. On the one hand, a node will contribute to a common goal based on cooperation without any assumption about previous knowledge of statistics or global observations. On the other hand, it applies a deep recurrent Q-network (DRQN) to address partial observability, improving the computational complexity.

In [84], it gets explained the Energy-Efficient Priority-based Multi-Objective QoS routing (PMQoSR) mechanism. It aims to ensure energy and QoS in IoT networks. The proposed system regulates the routing performance based on the QoS parameters, using an optimization technique for three hybrid algorithms (WLFA-Whale Lion Fireworks optimization algorithm with Fitness Function Routing mechanisms). The WLFA focuses on preventing congestion and minimizing localization errors by using and selecting the shortest path through the network.

In [85], an IoT-based proposal is outlined for monitoring Dengue disease. Data-based diagnosis is based on a Hybrid CNN with Tanh Long and Short-Term Memory (TLSTM) using the Adaptative Teaching Learning based Optimization algorithm.

In [86], the authors propose a routing protocol for IoT networks with low power consumption. A secure RPL (Routing Protocol for Low power Lossy network) protocol is introduced to work on congestion control in IoT environments. Authors describe 5 phases a) All nodes register in the sink node, which computes a hash based on the provided data (ID, IP, MAC, Rank, and PUF); b) It gets presented a Multi-context aware parent selection using fuzzy logic; c) The congestion detection is based on agents; d) A lightweight CNN is applied for deep packets analysis; e) An adaptative trickle timer works based on different parameters such as traffic intensity, etc.

In [87], the authors introduce a routing schedule method based on Timesensitive networking (TSN) for IIOT applications in underground mining. It discusses two alternatives to face the challenge associated with delays, walls, and sensor data collections a) a greedy algorithm based on the shortest distance, and b) a heuristic algorithm based on ant colony optimization.

In [88], the authors introduce a resource-efficient reliability model for MAS (Multi-Agent System) IoT systems for monitoring purposes. The proposed model starts with SLA requirements and characterizes it under the assumption of linear time complexity. It provides simplicity of computations and input metrics for aligning values in each range. Thus, it evaluates the time complexity, producing a measurement associated with the reliability of the model application.

In [89], the authors describe an identification method for rail vehicle running state using Tiny Machine Learning (TinyML). An IoT system is proposed focused on sizing and low energy consumption for data collection through Micro-Electro-Mechanical System (MEMS) sensors. The authors introduce a neural network (NN) model based on acceleration data for classifying the running states of rail vehicles. It represents a two-step approach where the model (i.e., NN) is trained after that is deployed on edge devices. An offset time window strategy addresses sensor results from edge devices and is uploaded to the cloud for visualization and analysis. In [90], it presents a unique IoT-based sensor node framework (aka, iThing). It aims to predict the onboard battery State of Health (SOH) and Remaining Useful Life (RUL) with the least computational and memory load. The SOH and RUL prediction are implemented through a random learning-based method using the voltage and health metrics of the device.

In [91], and in [5], it gets described as an edge analytics-assisted monitoring solution to monitor several physical activities of the patient. It focuses on determining physical inactivity from their daily routine. It wears wearable sensors to monitor physical movements and GPU for efficient data processing. The study pursues to calculate the scale of the physical inactivity of the patient to make real-time health suggestions.

In [92], it gets proposed how well-established data management techniques from the fields of computer science can be applied to data management for hydrological modeling. It is not a real-time approach but takes advantage of metadata modeling to enrich primary data from sensors. Data are complemented with spatial and temporal information that supports different decision-making models (e.g., upstream drainage area determination).

In [93], shown how the Gulf of Mexico Avian Monitoring Network members used structured decision-making to identify bird monitoring priorities by using multiple tools and techniques to clearly define the problem and stakeholder objectives and to identify bird monitoring priorities at the scale of the entire northern Gulf of Mexico region to coordinate the process of broad-scale monitoring programs to address management, restoration, and scientific questions in other ecosystems and for other taxa.

Study [94] focused on the use of big data in the decision-making processes of healthcare management, the abovementioned three aspects significantly affect the identification of management models, the distribution of responsibilities among professionals (therefore the evaluation of the partial results of each area responsibility) and the processes of allocating resources within the health-care organization.

6. Results and discussions. Independently of the year, citations, and decision-making strategy, this section focuses on answering every research question introduced in Section III and breaks down accordingly.

6.1. RQ1: Research Volume. The research volume in RTRM on EC is associated with fields like Networking (32%), Health (24%), Industrial (12%), Environment (10%), and Vehicle (8%), among others, as shown in Figure 3. Only those five fields represent 86% of the research volume and catch the scientific community's interest.

Figure 4 indicates a surface diagram contrasting the publishing year jointly with the citation volume and Scimago's quarter. As it is possible to appreciate, on the one hand, the intense research on the specific subject started to take an interest in 2013, with many citations focusing on Q1 journals. Even more, the volume of citations for recent years (e.g., 2021 and 2022) is close to the oldest works, which indicates an active development in the area. On the other hand, the citation over time shows an interesting growing perspective and opportunities in a context where the software and products reach (or propose) different As-a-Service strategies. Reliability is essential for achieving availability and scalability together [95].



Figure 5 shows that those results related to the research subject are published mainly in Q1 journals (70%), which are associated with a severe and strict publishing process jointly with a high interest in them from the scientific community and practitioners due to the quality of the content. It implies that the subject is worthy of addressing, while challenges and concerns are shared between stakeholders.



6.2. RQ2: System Reliability Monitoring Approaches. Figure 6 describes the proportion of data strategies associated with the AI models in reliability working on EC.



• Push-based • Pull-based • Distributed • Centralized Fig. 6. Proportion of Data Strategy Related to AI Models in Reliability

The heterogeneous and distributed context in which EC is deployed implies that data interoperability is mandatory. In this context, how data is collected from heterogeneous data sources is essential to reach the aim and make proper decisions based on the last known data. According to analyzed works, 76% of the systems use push-based strategies for RTRM, while 12% use a centralized approach. Only 10% address this challenge through a Pullbased process because it requires specific entity data under monitoring and specific communications requirements (not always available depending on the use case or scenario in which a solution is deployed). 2% introduced a distributed schema where data are local to the data source, and the interaction with the reliability component was guided through queries. Thus, the reliability component sends a query. At the same time, the answer is replied to using local data (with its limitations of it because the data are always partial to the data source).

Figure 7 describes the evolution of RTRM strategies over time. As it is possible to appreciate, it was focused on Push-based strategies between 2013 and 2019. In 2020, a centralized approach indicated a possible necessity of consolidation before making decisions around the system's reliability.

However, it additionally emerges Pull-based and distributed techniques between 2021 and 2022. In the end, there is no good or bad alternative because it would depend on the use case and kind of application that defines the specific monitoring requirements. It implies that a Pull-based approach could be interesting to watch distributed data centers because they could support Virtual Private Network (VPN) connections. Still, it would not be the best approach in Low-cost, tiny, distributed devices focusing on body sensor networks.



Fig. 7. Evolution of the Reliability Monitoring Strategies over Time

6.3. RQ3: Decision-making strategies. Even when edge computing is associated with a heterogeneous and distributed environment where data interoperability and data quality play an essential role, the main proportion of the system reliability decision-making approaches has been focused on centralized models from 2013 until now (Figure 8). It is a complex scenario because the ability to data analysis, make decisions, and act accordingly depends on a central component or element. Even when the central element could be redundant among other aspects, the figure of a central dependency is contradictory with the reliability perspective and a potential limitation to address the system scalability and high availability.



However, Federated and Collaborative decision-making approaches are taking interest from 2021 onwards. They propose a system in which the decision-making articulates alternative heterogeneous components monitoring different points of view in a distributed context and the possibility to collaborate, keeping a certain level of local autonomy.

The necessity of collaboration and avoiding central control has influenced Blockchain-based technologies significantly [96]. The idea is that the database content does not depend on a unique stakeholder; on the contrary, any addition or modification (or deletion) requires consensus among the participants. In the reliability context, it could be an exciting approach that the relative importance of each element in the complete system reliability could weigh.

6.4. RQ4: Data Collection Methods. Figure 9 describes the evolution of data collection methods associated with real-time monitoring systems. It was dominated by Direct and Hierarchical Push-based strategies between 2013 and 2020. Nevertheless, Direct-pull and Query-based approaches emerged as an alternative in 2021. 2022 is a context dominated by the Push approaches (be they hierarchical or not). It is consistent with the kind of areas in which the research volume is focusing, i.e., Networking (32%), Health (24%), Industrial (12%), Environment (10%), and Vehicles (8%).



An aspect worthy of mentioning is that independently of the increment of the research volume in this area, the push-based data collection approaches have become a central aspect to consider. On the other hand, those aspects related to data confidentiality, data sovereignty, and data ownership remain the main challenges and concerns.

6.5. RQ5: Applications. Figure 10 shows a tree map focusing on the main application areas dimensioned based on the citation volume, while inside each room, the most cited subjects are drawn per area.



Fig. 10. Main Subjects per Area based on the Citation Volume

In the networking field, the highest cited works are related to resource optimization, task scheduling, energy saving, and data accuracy. On the other hand, the health field focused on ECG telemetry systems, mHealth Reliability, arrhythmia detection, and patient monitoring.

An exciting work (worthy of reading and capitalizing) is introduced in the industrial field related to digital electronics in fibers. It gathers reliability, AI, and local data processing embedded in clothes fibers. Security authentication, machinery monitoring, and energy supply-demand matching are also addressed.

As it is logical, the vehicle field focused on vehicular monitoring, road status monitoring, and parking surveillance. Analogously, environmental areas focus on environmental and water monitoring due to its importance to people's health and relative impact.

Figure 11 outlines the evolution of application areas per year. It is not strange that the beginning is associated with environmental monitoring because it comes naturally from the IoT environment and the possibility to implement real-time monitoring systems at a relatively low cost [18].



Fig. 11. Application Areas per Year

However, the interest in the application on networking and health has been worthy of highlighting since 2018 onwards. The last three years have provided the context to extend the applications and increase the aspects to be analyzed under the reliability light.

6.6. General Considerations. After the analysis of the different research questions and the involved articles, it emerges essential aspects that are worth mentioning:

1. A few works mentioned using metadata to enrich sensor data to improve the data and automatic interpretation [81], [92]. It is essential to ensure semantic data interoperability because it directly knocks the heart of

system reliability due to the Data-driven decision-making approaches. Definitively, it is not a general approach; it constitutes an open subject, a challenge due to its complexity, and an opportunity to improve decision-making in qualitative terms.

2. The analyzed works need more measurement frameworks, or some formalization focused on guiding the experimental design jointly with the measurement process. It is critical because the strong assumption is that data used for system reliability monitoring are trustworthy, consistent (cohesive and coherent), accurate, precise, comparable, calibrated, and opportune. It is a current challenge and an opportunity to improve the current real-time system reliability monitoring systems.

3. Only [52] mentions the idea of calibration, but it is not associated with the data sources but with input features. The calibration process is essential in any device trying to quantify some property under monitoring because it ensures that the device is aligned with a given reference pattern. The system's entropy could affect (directly or not) the system components producing deviations due to the wear. Data-driven decision-making assumes that the consumed information is adequate for supporting the decision process. Calibration strategies in sensible areas like health and networking are an opportunity.

4. The process measurement assumptions are not reviewed. When data is received from different data sources, the decision-maker assumes that the data come from a repeatable measurement process (It is possible to obtain new data with a compatible method), results are comparable (It is possible to contrast numeric values over time), and extensible (It is possible to add a new characteristic to quantify). It represents an opportunity to integrate distributed and heterogeneous environments where reliability needs to be monitored in real-time.

5. Analyzed works discussed collecting and analyzing data for making decisions based on the system's aim. However, they do not add details about how to ensure the system's reliability and how to monitor it. That is to say; they do not say anything about how to ensure that the system requirements work as expected over time. It means that observability focuses on the typical telemetry and derived actions rather than how to ensure that the telemetry strategy and the system requirements are reached consistently over time. The meta-system monitoring needs to be addressed. For instance, observability is crucial for monitoring a certain level of availability for a given service. Still, also it is essential to monitor the process of producing the Service Level Agreement (SLA) that originated the instantiation of the offered service. The first refers to a particular SLA with a customer, but the second is associated with the matrix producing the offered SLA. Both must be monitored in terms of reliability.

6. In [69], the authors exposed the challenge of similar or redundant readings and the impact on data transmission latency and energy consumption. Semantic data interoperability is only sometimes considered, affecting how to optimize data retention and transmission policies from data sources (independent of the data collection methods). It is aggravated in data fusion scenarios because the potential integrations become relative and vague.

7. In [65], the authors described the data sparsity, incompleteness, and noise challenges due to the quick deployment of cheap sensors on the field. It is a good point because the IoT environment increases device density and coverage. Still, it needs to address the essential aspects of the data meaning and how to use them to support a decision-making process. Data quality aspects [17] could be analyzed more closely with system reliability.

As it is possible to appreciate, there are a lot of ongoing works on RTRM jointly with current concerns and opportunities. There is no standard approach around it, but there are multiple classes of proposals according to use cases or application areas. The subject has grown in importance in the last two years while leading international journals reported results that substantiate such growth.

7. Conclusion. This review showed the broad proposals and approaches of RTRM on EC. Most publications confirm the importance of the subject in areas such as Networking, Health, Industrial, Environment, and Vehicle. Nevertheless, there needs to be more consensus around monitoring reliability, ensuring syntactic and semantic data interoperability, formalizing the decision-making strategies in edge computing, and capitalizing on previous experiences and knowledge.

In terms of research volume, on the one hand, the intense research on the specific subject started to take an interest in 2013, where many citations focus on Q1 journals. On the other hand, the volume of citations for recent years (e.g., 2021, 2022 and 2023) is close to the oldest works, which indicates an active development in the area.

From the perspective of System Reliability Monitoring Approaches (SRMA): a) 76% of the systems use push-based strategies for reliability monitoring; b) 12% use a centralized approach; c) 10% address this challenge through a Pull-based approach; d) 2% introduced a distributed schema where data are local to the data source, and the interaction with the reliability component was guided through queries. The current dominant monitoring approach has been based on the Push strategy since 2013.

A centralized approach has dominated the decision-making strategies for reliability monitoring. However, the federated and collaborative
decision-making approaches are taking interest from 2021 onwards. They propose a system in which the decision-making articulates different heterogeneous components monitoring alternative points of view in a distributed context and the possibility to collaborate, keeping a certain level of local autonomy. The necessity of collaboration and avoiding central control has influenced Blockchain-based technologies in this area.

In terms of data collection methods, it is a clear leadership of Pushbased data collection methods (hierarchical or not). The non-hierarchical push-based methods represent 52.63%, while the hierarchical approach is associated with 31.58. Thus, in general, the participation of Puh-based data collection methods is 84.21%. The methods to use depend on the use cases and fields, among others. The remaining 15.79% is related to Pull-based data collection methods, composed of 5.26% for Direct Pull-based methods and 10.53% for Hierarchical Pull-based practices.

From the perspective of the application, Networking and Health areas represent 56% of published works. The percentage reaches 78% when the Industrial and Environment sectors are incorporated.

Beyond the research questions, there are essential aspects that need to be mentioned transversally as follows:

- A few works mentioned the use of metadata to enrich data coming from data sources to improve the data and automatic interpretation. Semantic data interoperability is an open opportunity in RTRM systems considering the distributed and heterogeneous environment associated with edge computing.

- Lack of measurement frameworks for formalizing a consistent, extensible, and repeatable measurement process. It represents a concern because it impacts the heart of Data-driven decision-making, but also an opportunity to improve reliability monitoring.

- Calibration is a subject analyzed on the surface in the works. However, it is essential to consider and maintain over time through different recalibration strategies. It is necessary to establish a reference pattern before obtaining measures. Also, this is a central resource when a metric is contrasted to decision criteria through indicators. Any data source (independently of the type) could fail or lose the calibration reference over time due to entropy.

- The underlying assumptions related to the measurement process are not analyzed. Neither is their impact on Data-driven decision-making. Thus, it represents an opportunity to improve reliability monitoring strategies through the online analysis of interest, comparability, accuracy, precision, trust, opportunity, coherence, cohesion, etc. It is a stage in which the strategy could discriminate among data, information, and knowledge. - The analyzed works focused on the aimed systems and how to collect and analyze data (e.g., patient monitoring). However, they rarely contemplated the meta-system monitoring to say whether (or not) system reliability is working as expected. Thus, meta-system analysis for monitoring different system properties (in addition to reliability) is open in this environment.

– Data ownership, Data sovereignty, Data Confidentiality, and Data Interoperability (Syntactic and Semantic) are rarely analyzed in the context of online reliability monitoring in EC. It represents a huge opportunity to make a difference. The security around the RTRM strategy is an open topic.

Thus, RTRM in EC is an ongoing work. It lacks standards but has taken importance and interest in the last two years. Articles focused on Push-based data collection methods for supporting centralized decisionmaking strategies. It was concentrated and deployed mainly on networking, health, industrial, and environmental monitoring. However, there are multiple opportunities and paths to walk to improve it. E.g., data interoperability, federated and collaborative decision-making models, formalization of the experimental design for measurement process, data sovereignty, organizational memory to capitalize previous knowledge (and experiences), calibration and recalibration strategies for data sources.

References

- Sun Y., Kadota I., Talak R., Modiano E. Age of Information: A New Metric for Information Freshness. Springer Cham, 2020. DOI: 10.2200/S00954ED2V01Y201909CNT023.
- Li H., Li X., Cheng Q. A fine-grained privacy protection data aggregation scheme for outsourcing smart grid. Frontiers of Computer Science. 2023. vol. 17. no. 3. DOI: 10.1007/s11704-022-2003-y.
- Murtadha M.K., Mushgil B.M. Flexible handover solution for vehicular ad-hoc networks based on software-defined networking and fog computing. International Journal of Electrical and Computer Engineering. 2023. vol. 13. no. 2. pp. 1570–1579. DOI: 10.11591/ijece.v13i2.
- Zhang H., Qi Q., Ji W., Tao F. An update method for digital twin multi-dimension models. Robotics and Computer-Integrated Manufacturing. 2023. vol. 80. DOI: 10.1016/j.rcim.2022.102481.
- 5. Algiriyage N., Prasanna R., Stock K., Doyle E.E., Johnston D. DEES: a real-time system for event extraction from disaster-related web text. Social Network Analysis and Mining. 2023. vol. 13. no. 1. DOI: 10.1007/s13278-022-01007-2.
- Mokhtar M.N.A.B.D., Ismail I., Hamzah W.M.A.F.W., Shamsuddin S.N.W., Arsad M.A.M. Real-Time Dream House Decorator in the Virtual Reality Environment. International Conference on Business and Technology. Cham: Springer International Publishing, 2021. vol. 487. pp. 525–537. DOI: 10.1007/978-3-031-08084-5_38.
- Grover J. Industrial IoT and Its Applications. IoT for Sustainable Smart Cities and Society. 2022. pp. 107–124. DOI: 10.1007/978-3-030-89554-9_5.

- Singh M., Srivastava R., Fuenmayor E., Kuts V., Qiao Y., Murray N., Devine D. Applications of Digital Twin across Industries: A Review. Applied Sciences. 2022. vol. 12. no. 11. DOI: 10.3390/app12115727.
- Wei C., Xu J., Li Q., Jiang S. An Intelligent Wildfire Detection Approach through Cameras Based on Deep Learning. Sustainability. 2022. vol. 14. no. 23. DOI: 10.3390/su142315690.
- Prabhu B.VB., Lakshmi R., Ankitha R., Prateeksha M.S., Priya N.C. RescueNet: YOLO-based object detection model for detection and counting of flood survivors. Modeling Earth Systems and Environment. 2022. vol. 8. no. 4. pp. 4509–4516. DOI: 10.1007/s40808-022-01414-6.
- Kumar N., Ramesh M.V. Accurate IoT Based Slope Instability Sensing System for Landslide Detection. IEEE Sensors Journal. 2022. vol. 22. no. 17. pp. 17151–17161. DOI: 10.1109/JSEN.2022.3189903.
- Wei Z., Zhu M., Zhang N., Wang L., Zou Y., Meng Z., Wu H., Feng Z. UAV-Assisted Data Collection for Internet of Things: A Survey. IEEE Internet of Things Journal. 2022. vol. 9. no. 17. pp. 15460–15483. DOI: 10.1109/JIOT.2022.3176903.
- Salau B.A., Rawal A., Rawat D.B. Recent Advances in Artificial Intelligence for Wireless Internet of Things and Cyber-Physical Systems: A Comprehensive Survey. IEEE Internet of Things Journal. 2022. vol. 9. no. 15. pp. 12916–12930. DOI: 10.1109/JIOT.2022.3170449.
- Costa B., Bachiega J., De Carvalho L.R., Araujo A.P. Orchestration in Fog Computing: A Comprehensive Survey. ACM Computing Surveys. 2023. vol. 55. no. 2. DOI: 10.1145/3486221.
- Niveditha P.S., John S.P., Simpson S.V. Review on Edge Computing-assisted d2d Networks. International Conference on Innovative Computing and Communications: Proceedings of ICICC. 2022. pp. 41–58. DOI: 10.1007/978-981-19-2821-5 4.
- Nguyen T.A., Kaliappan V.K., Jeon S., Jeon K.-S., Lee J.-W., Min D. Blockchain Empowered Federated Learning with Edge Computing for Digital Twin Systems in Urban Air Mobility. In Asia-Pacific International Symposium on Aerospace Technology. 2023. pp. 935–950. DOI: 10.1007/978-981-19-2635-8_69.
- ISO/IEC 25010. Systems and software engineering Systems and software Quality Requirements and Evaluation (SQuaRE) — System and software quality models. 2011. Available at: https://www.iso.org/standard/35733.html (accessed 20.09.2023).
- Divan M.J., Sanchez-Reynoso M.L., Panebianco J.E., Mendez M.J. IoT-Based Approaches for Monitoring the Particulate Matter and Its Impact on Health. IEEE Internet of Things Journal. 2021. vol. 8. no. 15. pp. 11983–12003. DOI: 10.1109/JIOT.2021.3068898.
- Divan M.J. Data-driven decision making. Proceeding of International Conference on Infocom Technologies and Unmanned Systems (Trends and Future Directions) (ICTUS). 2017. pp. 50–56. DOI: 10.1109/ICTUS.2017.8285973.
- Runeson P., Host M. Guidelines for conducting and reporting case study research in software engineering. Empirical software engineering. 2009. vol. 14. pp. 131–164. DOI: 10.1007/s10664-008-9102-8.
- Verner J.M., Sampson J., Tosic V., Bakar N.A., Kitchenham B.A. Guidelines for industrially-based multiple case studies in software engineering. Proceeding of Third International Conference on Research Challenges in Information Science. 2009. pp. 313–324. DOI: 10.1109/RCIS.2009.5089295.
- Petersen K., Vakkalanka S., Kuzniarz L. Guidelines for conducting systematic mapping studies in software engineering: An update. Information and software technology. 2015. vol. 64. pp. 1–18. DOI: 10.1016/j.infsof.2015.03.007.
- 23. Prajeesha, Anuradha M. EDGE Computing Application in SMART GRID-A Review. Proceeding of the 2nd International Conference on Electronics and Sustainable

Communication Systems (ICESC'2021). 2021. pp. 397–402. DOI: 10.1109/ICESC51422.2021.9532792.

- Anikwe C.V., Nweke H.F., Ikegwu A.C., Egwuonwu C.A., Onu F.U., Alo U.R., Teh Y.W. Mobile and wearable sensors for data-driven health monitoring system: State-ofthe-art and prospect. Expert Systems with Applications. 2022. vol. 202. DOI: 10.1016/j.eswa.2022.117362.
- Hu Z., Xu X., Zhang Y., Tang H., Cheng Y., Qian C., Khosravi M.R. Cloud–edge cooperation for meteorological radar big data: a review of data quality control. Complex and Intelligent Systems. 2021. DOI: 10.1007/s40747-021-00581-w.
- Groshev M., Guimaraes C., De La Oliva A., Gazda R. Dissecting the Impact of Information and Communication Technologies on Digital Twins as a Service. IEEE Access. 2021. vol. 9. pp. 102862–102876. DOI: 10.1109/ACCESS.2021.3098109.
- Begum B.A. Nandury S.V. A Survey of Data Aggregation Protocols for Energy Conservation in WSN and IoT. Wireless Communications and Mobile Computing. 2022. vol. 2022. DOI: 10.1155/2022/8765335.
- Gurewitz O., Shifrin M., Dvir E. Data Gathering Techniques in WSN: A Cross-Layer View. Sensors. 2022. vol. 22. no. 7. DOI: 10.3390/s22072650.
- Bayih A.Z., Morales J., Assabie Y., de By R.A. Utilization of Internet of Things and Wireless Sensor Networks for Sustainable Smallholder Agriculture. Sensors. 2022. vol. 22. no. 9. DOI: 10.3390/s22093273.
- Mirani A.A., Velasco-Hernandez G., Awasthi A., Walsh J. Key Challenges and Emerging Technologies in Industrial IoT Architectures: A Review. Sensors. 2022. vol. 22, no. 15. DOI: 10.3390/s22155836.
- Bagwari S., Gehlot A., Singh R., Priyadarshi N., Khan B. Low-Cost Sensor-Based and LoRaWAN Opportunities for Landslide Monitoring Systems on IoT Platform: A Review. IEEE Access. 2022. vol. 10. pp. 7107–7127. DOI: 10.1109/ACCESS.2021.3137841.
- Shahbazi Z., Byun Y.-C. Analysis of the Security and Reliability of Cryptocurrency Systems Using Knowledge Discovery and Machine Learning Methods. Sensors. 2022. vol. 22. no. 23. DOI: 10.3390/s22239083.
- Amador-Domínguez E., Serrano E., Manrique D. GEnI: A framework for the generation of explanations and insights of knowledge graph embedding predictions. Neurocomputing. 2023. vol. 521. pp. 199–212. DOI: 10.1016/j.neucom.2022.12.010.
- Valente F., Paredes S., Henriques J., Rocha T., de Carvalho P., Morais J. Interpretability, personalization and reliability of a machine learning based clinical decision support system. Data Min. Knowl. Discov. 2022. vol. 36. no. 3. pp. 1140– 1173. DOI: 10.1007/s10618-022-00821-8.
- Yang B., Bai X., Zhang C. Data Collection Method of Energy Adaptive Distributed Wireless Sensor Networks Based on UAV. Wirel. Commun. Mob. Comput. 2022. vol. 2022. DOI: 10.1155/2022/3469221.
- Bai Y., Cao L., Wang S., Ding H., Yue Y. Data Collection Strategy Based on OSELM and Gray Wolf Optimization Algorithm for Wireless Sensor Networks. Comput. Intell. Neurosci. 2022. vol. 2022. DOI: 10.1155/2022/4489436.
- Wei D. et al. Power-Efficient Data Collection Scheme for AUV-Assisted Magnetic Induction and Acoustic Hybrid Internet of Underwater Things. IEEE Internet Things J. 2022. vol. 9. no. 14. pp. 11675–11684. DOI: 10.1109/JIOT.2021.3131679.
- Benmansour F.L., Labraoui N.A. Comprehensive Review on Swarm Intelligence-Based Routing Protocols in Wireless Multimedia Sensor Networks. Int. J. Wirel. Inf. Networks. 2021. vol. 28. no. 2. pp. 175–198. DOI: 10.1007/s10776-021-00508-9.
- Hannan M.A., Hassan K., Jern K.P. A review on sensors and systems in structural health monitoring: Current issues and challenges. Smart Struct. Syst. 2018. vol. 22. no. 5. pp. 509–525. DOI: 10.12989/sss.2018.22.5.509.

- Ajakwe S.O., Nwakanma C.I., Kim D.-S., Lee J.-M. Key Wearable Device Technologies Parameters for Innovative Healthcare Delivery in B5G Network: A Review. IEEE Access. 2022. vol. 10. pp. 49956–49974. DOI: 10.1109/ACCESS.2022.3173643.
- SJR SCImago Journal and Country Rank [Official web site of SCImago]. 2020. Available at: www.scimagojr.com (accessed 23.09.2020).
- Guerrero-Bote V.P., Moya-Anegón F. A further step forward in measuring journals' scientific prestige: The SJR2 indicator. J. Informetr. 2012. vol. 6. no. 4. pp. 674–688. DOI: 10.1016/j.joi.2012.07.001.
- Sung W., Hsu C. IOT system environmental monitoring using IPSO weight factor estimation. Sens. Rev. 2013. vol. 33. no. 3, pp. 246–256. DOI: 10.1108/02602281311324708.
- 44. Pournaras E., Yao M., Helbing D. Self-regulating supply-demand systems. Futur. Gener. Comput. Syst. 2017. vol. 76. pp. 73–91. DOI: 10.1016/j.future.2017.05.018.
- Satija U., Ramkumar B., Manikandan M.S. Real-Time Signal Quality-Aware ECG Telemetry System for IoT-Based Health Care Monitoring. IEEE Internet Things J. 2017. vol. 4. no. 3. pp. 815–823. DOI: 10.1109/JIOT.2017.2670022.
- Al-Jaroodi J., Mohamed N. PsCPS: A Distributed Platform for Cloud and Fog Integrated Smart Cyber-Physical Systems. IEEE Access. 2018. vol. 6. pp. 41432– 41449. DOI: 10.1109/ACCESS.2018.2856509.
- 47. Pore M., Chakati V., Banerjee A., Gupta S.K.S. ContextAiDe. ACM Trans. Internet Technol. 2019. vol. 19. no. 2. pp. 1–23. DOI: 10.1145/3301444.
- Albahri O.S. et al. Fault-Tolerant mHealth Framework in the Context of IoT-Based Real-Time Wearable Health Data Sensors. IEEE Access. 2019. vol. 7, pp. 50052– 50080. DOI: 10.1109/ACCESS.2019.2910411.
- Cao K., Xu G., Zhou J., Wei T., Chen M., Hu S. QoS-Adaptive Approximate Real-Time Computation for Mobility-Aware IoT Lifetime Optimization. IEEE Trans. Comput. Des. Integr. Circuits Syst. 2019. vol. 38. no. 10. pp. 1799–1810. DOI: 10.1109/TCAD.2018.2873239.
- Dong R., She C., Hardjawana W., Li Y., Vucetic B. Deep Learning for Hybrid 5G Services in Mobile Edge Computing Systems: Learn From a Digital Twin. IEEE Trans. Wirel. Commun. 2019. vol. 18. no. 10. pp. 4692–4707. DOI: 10.1109/TWC.2019.2927312.
- Xu X., He C., Xu Z., Qi L., Wan S., Bhuiyan M.Z.A. Joint Optimization of Offloading Utility and Privacy for Edge Computing Enabled IoT. IEEE Internet Things J. 2020. vol. 7. no. 4. pp. 2622–2629. DOI: 10.1109/JIOT.2019.2944007.
- Farahani B., Barzegari M., Aliee F.S., Shaik K.A. Towards collaborative intelligent IoT eHealth: From device to fog, and cloud. Microprocess. Microsyst. 2020. vol. 72. p. 102938. DOI: 10.1016/j.micpro.2019.102938.
- Zhang T. et al. A Joint Deep Learning and Internet of Medical Things Driven Framework for Elderly Patients. IEEE Access. 2020. vol. 8. pp. 75822–75832. DOI: 10.1109/ACCESS.2020.2989143.
- Sodhro A.H., Sodhro G.H., Guizani M., Pirbhulal S., Boukerche A. AI-Enabled Reliable Channel Modeling Architecture for Fog Computing Vehicular Networks. IEEE Wirel. Commun. 2020. vol. 27. no. 2. pp. 14–21. DOI: 10.1109/MWC.001.1900311.
- Bhatia M., Sood S.K. Quantum Computing-Inspired Network Optimization for IoT Applications. IEEE Internet Things J. 2020. vol. 7. no. 6. pp. 5590–5598. DOI: 10.1109/JIOT.2020.2979887.
- Zhang J. Research on environmental monitoring trend analysis based on internet of things visualization technology. Fresenius Environ. Bull. 2020. vol. 29. no. 2. pp. 1054 – 1062. [Online]. Available at: www.scopus.com/inward/record.uri?eid=2-s2.0-

85090455681&partnerID=40&md5=7c5626045c3f39320bb91185bf694295 (accessed 11.05.2023).

- Loke G. et al. Digital electronics in fibers enable fabric-based machine-learning inference. Nat. Commun. 2021. vol. 12. no. 1. p. 3317. DOI: 10.1038/s41467-021-23628-5.
- Ke R., Zhuang Y., Pu Z., Wang Y. A Smart, Efficient, and Reliable Parking Surveillance System with Edge Artificial Intelligence on IoT Devices. IEEE Trans. Intell. Transp. Syst. 2021. vol. 22. no. 8. pp. 4962–4974. DOI: 10.1109/TITS.2020.2984197.
- Wang X., Garg S., Lin H., Piran M.J., Hu J., Hossain M.S. Enabling Secure Authentication in Industrial IoT with Transfer Learning Empowered Blockchain. IEEE Trans. Ind. Informatics. 2021. vol. 17. no. 11. pp. 7725–7733. DOI: 10.1109/TII.2021.3049405.
- Lee W.J., Xia K., Denton N.L., Ribeiro B., Sutherland J.W. Development of a speed invariant deep learning model with application to condition monitoring of rotating machinery. J. Intell. Manuf. 2021. vol. 32. no. 2. pp. 393–406. DOI: 10.1007/s10845-020-01578-x.
- Ibrar M., Wang L., Muntean G.-M., Chen J., Shah N., Akbar A. IHSF: An Intelligent Solution for Improved Performance of Reliable and Time-Sensitive Flows in Hybrid SDN-Based FC IoT Systems. IEEE Internet Things J. 2021. vol. 8. no. 5. pp. 3130– 3142. DOI: 10.1109/JIOT.2020.3024560.
- Brik B., Esseghir M., Merghem-Boulahia L., Snoussi H. An IoT-based deep learning approach to analyse indoor thermal comfort of disabled people. Build. Environ. 2021. vol. 203. p. 108056. DOI: 10.1016/j.buildenv.2021.108056.
- Singh P.D., Kaur R., Singh K.D., Dhiman G., Soni M. Fog-centric IoT based smart healthcare support service for monitoring and controlling an epidemic of Swine Flu virus. Informatics Med. Unlocked. 2021. vol. 26. p. 100636. DOI: 10.1016/j.imu.2021.100636.
- Bhatia M. Intelligent System of Game-Theory-Based Decision Making in Smart Sports Industry. ACM Trans. Intell. Syst. Technol. 2021. vol. 12. no. 3. pp. 1–23. DOI: 10.1145/3447986.
- Baggag A. et al. Learning Spatiotemporal Latent Factors of Traffic via Regularized Tensor Factorization: Imputing Missing Values and Forecasting. IEEE Trans. Knowl. Data Eng. 2021. vol. 33. no. 6. pp. 2573–2587. DOI: 10.1109/TKDE.2019.2954868.
- Razzaq M.A., Mahar J.A., Ahmad M., Saher N., Mehmood A., Choi G.S. Hybrid Auto-Scaled Service-Cloud-Based Predictive Workload Modeling and Analysis for Smart Campus System. IEEE Access. 2021. vol. 9. pp. 42081–42089. DOI: 10.1109/ACCESS.2021.3065597.
- 67. Jin H., Zhao J. Real-time energy consumption detection simulation of network node in internet of things based on artificial intelligence. Sustainable Energy Technologies and Assessments. 2021. vol. 44. no. 101004. DOI: 10.1016/j.seta.2021.101004.
- Alzamzami F., El Saddik A. Monitoring Cyber SentiHate Social Behavior During COVID-19 Pandemic in North America. IEEE Access. 2021. vol. 9. pp. 91184– 91208. DOI: 10.1109/ACCESS.2021.3088410.
- Adhikari M., Ambigavathi M., Menon V.G., Hammoudeh M. Random Forest for Data Aggregation to Monitor and Predict COVID-19 Using Edge Networks. IEEE Internet Things Mag. 2021. vol. 4. no. 2. pp. 40–44. DOI: 10.1109/IOTM.0001.2100052.
- Jurdi R., Andrews J.G., Heath R.W. Scheduling Observers Over a Shared Channel With Hard Delivery Deadlines. IEEE Trans. Commun. 2021. vol. 69. no. 1. pp. 133– 148. DOI: 10.1109/TCOMM.2020.3032172.
- 71. Hashash O., Sharafeddine S., Dawy Z., Mohamed A., Yaacoub E. Energy-Aware Distributed Edge ML for mHealth Applications with Strict Latency Requirements.
- 1284 Информатика и автоматизация. 2023. Том 22 № 6. ISSN 2713-3192 (печ.) ISSN 2713-3206 (онлайн) www.ia.spcras.ru

IEEE Wirel. Commun. Lett. 2021. vol. 10. no. 12. pp. 2791–2794. DOI: 10.1109/LWC.2021.3117876.

- Liang W., Li W., Feng L. Information Security Monitoring and Management Method Based on Big Data in the Internet of Things Environment. IEEE Access. 2021. vol. 9. pp. 39798–39812. DOI: 10.1109/ACCESS.2021.3064350.
- Shao S., Zhang Q., Guo S., Qi F. Task Allocation Mechanism for Cable Real-Time Online Monitoring Business Based on Edge Computing. IEEE Syst. J. 2021. vol. 15. no. 1. pp. 1344–1355. DOI: 10.1109/JSYST.2020.2988266.
- Vaidya G., Nambi A., Prabhakar T.V., Kumar V.T., Sudhakara S. Towards generating a reliable device-specific identifier for IoT devices. Pervasive Mob. Comput. 2021. vol. 76. DOI: 10.1016/j.pmcj.2021.101445.
- Rajendran S. et al. Emphasizing privacy and security of edge intelligence with machine learning for healthcare. Int. J. Intell. Comput. Cybern. 2022. vol. 15. no. 1. pp. 92–109. DOI: 10.1108/IJICC-05-2021-0099.
- Zahid N., Sodhro A.H., Kamboh U.R., Alkhayyat A., Wang L. AI-driven adaptive reliable and sustainable approach for internet of things enabled healthcare system. Math. Biosci. Eng. 2022. vol. 19. no. 4. pp. 3953–3971. DOI: 10.3934/mbe.2022182.
- Bhardwaj A. et al. Smart IoT and Machine Learning-based Framework for Water Quality Assessment and Device Component Monitoring. Environ. Sci. Pollut. Res. 2022. vol. 29. no. 30. pp. 46018–46036. DOI: 10.1007/s11356-022-19014-3.
- Alsalemi A., Himeur Y., Bensaali F., Amira A. An innovative edge-based Internet of Energy solution for promoting energy saving in buildings. Sustainable Cities and Society. 2022. vol. 78. no. 103571. DOI: 10.1016/j.scs.2021.103571.
- Khan M.A., Ghazal T.M., Lee S.-W., Rehman A. Data Fusion-Based Machine Learning Architecture for Intrusion Detection. Comput. Mater. Contin. 2022. vol. 70. no. 2. pp. 3399–3413. DOI: 10.32604/cmc.2022.020173.
- Gültekin Ö., Cinar E., Özkan K., Yazıcı A. Real-Time Fault Detection and Condition Monitoring for Industrial Autonomous Transfer Vehicles Utilizing Edge Artificial Intelligence. Sensors. 2022. vol. 229(9). no. 3208. DOI: 10.3390/s22093208.
- Wei L., Hou S., Liu Q. Clinical Care of Hyperthyroidism Using Wearable Medical Devices in a Medical IoT Scenario. Journal of Healthcare Engineering. 2022. vol. 2022. DOI: 10.1155/2022/5951326.
- Nikolov G., Kuhn M., Mcgibney A., Wenning B.-L. MABASR A Robust Wireless Interface Selection Policy for Heterogeneous Vehicular Networks. IEEE Access. 2022. vol. 10. pp. 26068–26077. DOI: 10.1109/ACCESS.2022.3156597.
- Baek J., Kaddoum G. Online Partial Offloading and Task Scheduling in SDN-Fog Networks With Deep Recurrent Reinforcement Learning. IEEE Internet Things J. 2022. vol. 9. no. 13. pp. 11578–11589. DOI: 10.1109/JIOT.2021.3130474.
- Thenmozhi R., Sakthivel P., Kulothungan K. Hybrid multi-objective-optimization algorithm for energy efficient priority-based QoS routing in IoT networks. Wireless Networks. 2022. DOI: 10.1007/s11276-021-02848-z.
- Manoharan S.N., Kumar K.M.V.M., Vadivelan N.A. Novel CNN-TLSTM Approach for Dengue Disease Identification and Prevention using IoT-Fog Cloud Architecture. Neural Processing Letters. 2022. vol. 55. no. 2. pp. 1951–1973. DOI: 10.1007/s11063-022-10971-x.
- Sithik M.M., Kumar B.M. Intelligent agent based virtual clustering and multi-context aware routing for congestion mitigation in secure RPL-IoT environment. Ad Hoc Networks. 2022. vol. 137. DOI: 10.1016/j.adhoc.2022.102972.
- Zhang Y., Wu J., Liu M., Tan A. TSN-based routing and scheduling scheme for Industrial Internet of Things in underground mining. Engineering Applications of Artificial Intelligence. 2022. vol. 115. DOI: 10.1016/j.engappai.2022.105314.

- Eroshkin I., Vojtech L., Neruda M. Resource Efficient Real-Time Reliability Model for Multi-Agent IoT Systems. IEEE Access. 2022. vol. 10. pp. 2578–2590. DOI: 10.1109/ACCESS.2021.3138931.
- Zhou S., Du Y., Chen B., Li Y., Luan X. An Intelligent IoT Sensing System for Rail Vehicle Running States Based on TinyML. IEEE Access. 2022. vol. 10. pp. 98860– 98871. DOI: 10.1109/ACCESS.2022.3206954.
- Sinha A., Das D., Udutalapally V., Mohanty S.P. iThing: Designing Next-Generation Things with Battery Health Self-Monitoring Capabilities for Sustainable IIoT. IEEE Trans. Instrum. Meas. 2022. vol. 71. pp. 1–9. DOI: 10.1109/TIM.2022.3216594.
- Manocha A., Singh R. A Novel Edge Analytics Assisted Motor Movement Recognition Framework Using Multi-Stage Convo-GRU Model. Mob. Networks Appl. 2022. vol. 27. no. 2. pp. 657–676. DOI: 10.1007/s11036-019-01321-8.
- Bollen E. et al. A database system for querying of river networks: facilitating monitoring and prediction applications. Water Supply. 2022. vol. 22. no. 3. pp. 2832– 2846. DOI: 10.2166/ws.2021.433.
- Fournier A.M.V. et al. Structured Decision Making to Prioritize Regional Bird Monitoring Needs. INFORMS Journal on Applied Analytics. 2023. vol. 53(3). pp. 207–217. DOI: 10.1287/inte.2022.1154.
- Fanelli S., Pratici L., Salvatore F.P., Donelli C.C., Zangrandi A. Big data analysis for decision-making processes: challenges and opportunities for the management of health-care organizations. Management Research Review. 2023. vol. 46. no. 3. pp. 369–389. DOI: 10.1108/MRR-09-2021-0648.
- Overdal M., Haddara M., Langseth M. Exploring Public Cloud-ERP Systems' Impact on Organizational Performance. Lect. Notes Networks Syst. 2023. vol. 561. pp. 121– 137. DOI: 10.1007/978-3-031-18344-7_8.
- Reyna A., Martín C., Chen J., Soler E., Díaz M. On blockchain and its integration with IoT. Challenges and opportunities. Futur. Gener. Comput. Syst. 2018. vol. 88. pp. 173–190. DOI: 10.1016/j.future.2018.05.046.

Diván Mario José — Ph.D., Honored professor, Sr. solutions architect, Department of edge reliability engineering, Intel Corporation. Research interests: cyber security, management, scalability and monetization of IT service platforms. The number of publications — 91. mario.jose.divan.koller@intel.com; 3, 25th Ave., Jones Farm Campus, 97124, Hillsboro, Oregon, United States of America; office phone: +1(503)264-0180.

Shchemelinin Dmitry — Ph.D., Dr.Sci., Head (lead) of the department, Department of cloud site reliability engineering, Intel Corporation. Research interests: monitoring, state management and scalability of globally distributed computing systems by using methods of artificial intelligence, big data collection and analysis. The number of publications — 78. dshchmel@gmail.com; 3, 25th Ave., Jones Farm Campus, 97124, Hillsboro, Oregon, United States of America; office phone: +1(408)981-5963.

Carranza Marcos E. — Principal engineer, Department of of edge reliability engineering, Intel Corporation. Research interests: cyber security, management, scalability and monetization of IT service platforms. The number of publications — 32. marcos.e.carranza@intel.com; 3, 25th Ave., Jones Farm Campus, 97124, Hillsboro, Oregon, United States of America; office phone: +1(503)764-7952.

Martinez-Spessot Cesar Ignacio — Head (lead) of the department, Department of edge reliability engineering, Intel Corporation. Research interests: application of digital technologies by using artificial intelligence methods, development of the internet of things ecosystems and applications. The number of publications — 31. cesar.martinez@intel.com; 3, 25th Ave., Jones

Farm Campus, 97124, Hillsboro, Oregon, United States of America; office phone: +1(503)7120559.

Buinevich Mikhail — Ph.D., Dr.Sci., Professor of the department, Department of applied mathematics and information technologies, Saint-Petersburg University of State Fire Service of EMERCOM of Russia. Research interests: cyber antifragility of info telecommunications systems, artificial intelligence methods in processing data state of complex systems, training of specialists in information security, cyber polygon. The number of publications — 170. bmv1958@yandex.ru; 149, Moskovskij Ave., 196105, St. Petersburg, Russia; office phone: +7(812)388-8639.

Acknowledgements. This work was fully supported by Intel Corporation.

УДК 004.75

DOI 10.15622/ia.22.6.1

М.Х. Диван, Д.А. Щемелинин, М. Карранса, Ц.И. Мартинес-Спессот, М.В. Буйневич МОНИТОРИНГ НАДЕЖНОСТИ ПОЛЬЗОВАТЕЛЬСКИХ ВЫЧИСЛИТЕЛЬНЫХ УСТРОЙСТВ В РЕЖИМЕ РЕАЛЬНОГО ВРЕМЕНИ: СИСТЕМАТИЧЕСКОЕ ОТОБРАЖЕНИЕ

Диван М.Х., Щемелинин Д.А., Карранса М., Мартинес-Спессот Ц.И., Буйневич М.В. Мониторинг надежности пользовательских вычислительных устройств в режиме реального времени: систематическое отображение.

Аннотация. Данный исследовательский обзор сосредоточен на мониторинге надежности вычислительных систем в режиме реального времени на стороне пользователя. В условиях гетерогенной и распределенной вычислительной среды, где отсутствует централизованный контроль, исследуется использование моделей искусственного интеллекта для поддержки процессов принятия решений в мониторинге надежности системы. Методология исследования основана на систематическом отображении предыдущих исследований, опубликованных в научных базах данных IEEE и Scopus. Анализ проведен на основе 50 научных статей, опубликованных с 2013 по 2022 годы, показал растущий научный интерес к данной области. Основное применение исследуемого метода связано с сетевыми технологиями И здравоохранением. Данный метод нацелен на интеграцию сети медицинских сенсоров и управляющих данных с пользовательскими вычислительными устройствами. Однако этот метод также применяется в промышленном и экологическом мониторинге. Выводы что исследования показывают. мониторинг належности пользовательских вычислительных устройств в режиме реального времени находится на начальной стадии развития. Он не имеет стандартов, но за последние два года приобрел значительное значение и интерес. Большинство исследуемых статей сосредоточены на методах сбора данных с использованием уведомлений для поддержки централизованных стратегий принятия решений. Однако, существует множество возможностей для дальнейшего развития данного метода, таких как совместимость данных, федеративные и совместные модели принятия решений, формализация экспериментального дизайна, суверенитет данных, систематизация базы данных для использования предыдущих знаний и опыта, стратегии калибровки и повторной корректировки для источников данных.

Ключевые слова: реальное время, надежность системы, мониторинг, вычисления на стороне пользователя, систематическое исследование.

Литература

- Sun Y., Kadota I., Talak R., Modiano E. Age of Information: A New Metric for Information Freshness // Springer Cham, 2020. DOI: 10.2200/S00954ED2V01Y201909CNT023.
- Li H., Li X., Cheng Q. A fine-grained privacy protection data aggregation scheme for outsourcing smart grid // Frontiers of Computer Science. 2023. vol. 17. no. 3. DOI: 10.1007/s11704-022-2003-y.
- Murtadha M.K., Mushgil B.M. Flexible handover solution for vehicular ad-hoc networks based on software-defined networking and fog computing // International Journal of Electrical and Computer Engineering. 2023. vol. 13. no. 2. pp. 1570–1579. DOI: 10.11591/ijece.v13i2.

- Zhang H., Qi Q., Ji W., Tao F. An update method for digital twin multi-dimension models // Robotics and Computer-Integrated Manufacturing. 2023. vol. 80. DOI: 10.1016/j.rcim.2022.102481.
- Algiriyage N., Prasanna R., Stock K., Doyle E.E., Johnston D. DEES: a real-time system for event extraction from disaster-related web text // Social Network Analysis and Mining. 2023. vol. 13. no. 1. DOI: 10.1007/s13278-022-01007-2.
- Mokhtar M.N.A.B.D., Ismail I., Hamzah W.M.A.F.W., Shamsuddin S.N.W., Arsad M.A.M. Real-Time Dream House Decorator in the Virtual Reality Environment // International Conference on Business and Technology. Cham: Springer International Publishing, 2021. vol. 487. pp. 525–537. DOI: 10.1007/978-3-031-08084-5_38.
- Grover J. Industrial IoT and Its Applications // IoT for Sustainable Smart Cities and Society. 2022. pp. 107–124. DOI: 10.1007/978-3-030-89554-9 5.
- Singh M., Srivastava R., Fuenmayor E., Kuts V., Qiao Y., Murray N., Devine D. Applications of Digital Twin across Industries: A Review // Applied Sciences. 2022. vol. 12. no. 11. DOI: 10.3390/app12115727.
- Wei C., Xu J., Li Q., Jiang S. An Intelligent Wildfire Detection Approach through Cameras Based on Deep Learning // Sustainability. 2022. vol. 14. no. 23. DOI: 10.3390/su142315690.
- Prabhu B.VB., Lakshmi R., Ankitha R., Prateeksha M.S., Priya N.C. RescueNet: YOLO-based object detection model for detection and counting of flood survivors // Modeling Earth Systems and Environment. 2022. vol. 8. no. 4. pp. 4509–4516. DOI: 10.1007/s40808-022-01414-6.
- Kumar N., Ramesh M.V. Accurate IoT Based Slope Instability Sensing System for Landslide Detection // IEEE Sensors Journal. 2022. vol. 22. no. 17. pp. 17151–17161. DOI: 10.1109/JSEN.2022.3189903.
- Wei Z., Zhu M., Zhang N., Wang L., Zou Y., Meng Z., Wu H., Feng Z. UAV-Assisted Data Collection for Internet of Things: A Survey // IEEE Internet of Things Journal. 2022. vol. 9. no. 17. pp. 15460–15483. DOI: 10.1109/JIOT.2022.3176903.
- Salau B.A., Rawal A., Rawat D.B. Recent Advances in Artificial Intelligence for Wireless Internet of Things and Cyber-Physical Systems: A Comprehensive Survey // IEEE Internet of Things Journal. 2022. vol. 9. no. 15. pp. 12916–12930. DOI: 10.1109/JIOT.2022.3170449.
- Costa B., Bachiega J., De Carvalho L.R., Araujo A.P. Orchestration in Fog Computing: A Comprehensive Survey // ACM Computing Surveys. 2023. vol. 55. no. 2. DOI: 10.1145/3486221.
- Niveditha P.S., John S.P., Simpson S.V. Review on Edge Computing-assisted d2d Networks // International Conference on Innovative Computing and Communications: Proceedings of ICICC. 2022. pp. 41–58. DOI: 10.1007/978-981-19-2821-5_4.
- Nguyen T.A., Kaliappan V.K., Jeon S., Jeon K.-S., Lee J.-W., Min D. Blockchain Empowered Federated Learning with Edge Computing for Digital Twin Systems in Urban Air Mobility // In Asia-Pacific International Symposium on Aerospace Technology. 2023. pp. 935–950. DOI: 10.1007/978-981-19-2635-8 69.
- ISO/IEC 25010. Systems and software engineering Systems and software Quality Requirements and Evaluation (SQuaRE) — System and software quality models. 2011. Available at: https://www.iso.org/standard/35733.html (accessed 20.09.2023).
- Divan M.J., Sanchez-Reynoso M.L., Panebianco J.E., Mendez M.J. IoT-Based Approaches for Monitoring the Particulate Matter and Its Impact on Health // IEEE Internet of Things Journal. 2021. vol. 8. no. 15. pp. 11983–12003. DOI: 10.1109/JIOT.2021.3068898.
- Divan M.J. Data-driven decision making // Proceeding of International Conference on Infocom Technologies and Unmanned Systems (Trends and Future Directions) (ICTUS). 2017. pp. 50–56. DOI: 10.1109/ICTUS.2017.8285973.

- Runeson P., Host M. Guidelines for conducting and reporting case study research in software engineering // Empirical software engineering. 2009. vol. 14. pp. 131–164. DOI: 10.1007/s10664-008-9102-8.
- Verner J.M., Sampson J., Tosic V., Bakar N.A., Kitchenham B.A. Guidelines for industrially-based multiple case studies in software engineering // Proceeding of Third International Conference on Research Challenges in Information Science. 2009. pp. 313–324. DOI: 10.1109/RCIS.2009.5089295.
- Petersen K., Vakkalanka S., Kuzniarz L. Guidelines for conducting systematic mapping studies in software engineering: An update // Information and software technology. 2015. vol. 64. pp. 1–18. DOI: 10.1016/j.infsof.2015.03.007.
- Prajeesha, Anuradha M. EDGE Computing Application in SMART GRID-A Review // Proceeding of the 2nd International Conference on Electronics and Sustainable Communication Systems (ICESC'2021). 2021. pp. 397–402. DOI: 10.1109/ICESC51422.2021.9532792.
- Anikwe C.V., Nweke H.F., Ikegwu A.C., Egwuonwu C.A., Onu F.U., Alo U.R., Teh Y.W. Mobile and wearable sensors for data-driven health monitoring system: State-ofthe-art and prospect // Expert Systems with Applications. 2022. vol. 202. DOI: 10.1016/j.eswa.2022.117362.
- Hu Z., Xu X., Zhang Y., Tang H., Cheng Y., Qian C., Khosravi M.R.. Cloud–edge cooperation for meteorological radar big data: a review of data quality control // Complex and Intelligent Systems. 2021. DOI: 10.1007/s40747-021-00581-w.
- Groshev M., Guimaraes C., De La Oliva A., Gazda R. Dissecting the Impact of Information and Communication Technologies on Digital Twins as a Service // IEEE Access. 2021. vol. 9. pp. 102862–102876. DOI: 10.1109/ACCESS.2021.3098109.
- Begum B.A. Nandury S.V. A Survey of Data Aggregation Protocols for Energy Conservation in WSN and IoT // Wireless Communications and Mobile Computing. 2022. vol. 2022. DOI: 10.1155/2022/8765335.
- Gurewitz O., Shifrin M., Dvir E. Data Gathering Techniques in WSN: A Cross-Layer View // Sensors. 2022. vol. 22. no. 7. DOI: 10.3390/s22072650.
- Bayih A.Z., Morales J., Assabie Y., de By R.A. Utilization of Internet of Things and Wireless Sensor Networks for Sustainable Smallholder Agriculture // Sensors. 2022. vol. 22. no. 9. DOI: 10.3390/s22093273.
- Mirani A.A., Velasco-Hernandez G., Awasthi A., Walsh J. Key Challenges and Emerging Technologies in Industrial IoT Architectures: A Review // Sensors. 2022. vol. 22, no. 15. DOI: 10.3390/s22155836.
- Bagwari S., Gehlot A., Singh R., Priyadarshi N., Khan B. Low-Cost Sensor-Based and LoRaWAN Opportunities for Landslide Monitoring Systems on IoT Platform: A Review // IEEE Access. 2022. vol. 10. pp. 7107–7127. DOI: 10.1109/ACCESS.2021.3137841.
- Shahbazi Z., Byun Y.-C. Analysis of the Security and Reliability of Cryptocurrency Systems Using Knowledge Discovery and Machine Learning Methods // Sensors. 2022. vol. 22. no. 23. DOI: 10.3390/s22239083.
- Amador-Domínguez E., Serrano E., Manrique D. GEnI: A framework for the generation of explanations and insights of knowledge graph embedding predictions // Neurocomputing. 2023. vol. 521. pp. 199–212. DOI: 10.1016/j.neucom.2022.12.010.
- Valente F., Paredes S., Henriques J., Rocha T., de Carvalho P., Morais J. Interpretability, personalization and reliability of a machine learning based clinical decision support system // Data Min. Knowl. Discov. 2022. vol. 36. no. 3. pp. 1140– 1173. DOI: 10.1007/s10618-022-00821-8.
- Yang B., Bai X., Zhang C. Data Collection Method of Energy Adaptive Distributed Wireless Sensor Networks Based on UAV // Wirel. Commun. Mob. Comput. 2022. vol. 2022. DOI: 10.1155/2022/3469221.
- 1290 Информатика и автоматизация. 2023. Том 22 № 6. ISSN 2713-3192 (печ.) ISSN 2713-3206 (онлайн) www.ia.spcras.ru

- Bai Y., Cao L., Wang S., Ding H., Yue Y. Data Collection Strategy Based on OSELM and Gray Wolf Optimization Algorithm for Wireless Sensor Networks // Comput. Intell. Neurosci. 2022. vol. 2022. DOI: 10.1155/2022/4489436.
- Wei D. et al. Power-Efficient Data Collection Scheme for AUV-Assisted Magnetic Induction and Acoustic Hybrid Internet of Underwater Things // IEEE Internet Things J. 2022. vol. 9. no. 14. pp. 11675–11684. DOI: 10.1109/JIOT.2021.3131679.
- Benmansour F.L., Labraoui N.A. Comprehensive Review on Swarm Intelligence-Based Routing Protocols in Wireless Multimedia Sensor Networks // Int. J. Wirel. Inf. Networks. 2021. vol. 28. no. 2. pp. 175–198. DOI: 10.1007/s10776-021-00508-9.
- Hannan M.A., Hassan K., Jern K.P. A review on sensors and systems in structural health monitoring: Current issues and challenges // Smart Struct. Syst. 2018. vol. 22. no. 5. pp. 509–525. DOI: 10.12989/sss.2018.22.5.509.
- Ajakwe S.O., Nwakanma C.I., Kim D.-S., Lee J.-M. Key Wearable Device Technologies Parameters for Innovative Healthcare Delivery in B5G Network: A Review // IEEE Access. 2022. vol. 10. pp. 49956–49974. DOI: 10.1109/ACCESS.2022.3173643.
- SJR SCImago Journal and Country Rank [Official web site of SCImago]. 2020. Available at: www.scimagojr.com (accessed 23.09.2020).
- Guerrero-Bote V.P., Moya-Anegón F. A further step forward in measuring journals' scientific prestige: The SJR2 indicator // J. Informetr. 2012. vol. 6. no. 4. pp. 674–688. DOI: 10.1016/j.joi.2012.07.001.
- Sung W., Hsu C. IOT system environmental monitoring using IPSO weight factor estimation // Sens. Rev. 2013. vol. 33. no. 3, pp. 246–256. DOI: 10.1108/02602281311324708.
- Pournaras E., Yao M., Helbing D. Self-regulating supply-demand systems // Futur. Gener. Comput. Syst. 2017. vol. 76. pp. 73–91. DOI: 10.1016/j.future.2017.05.018.
- Satija U., Ramkumar B., Manikandan M.S. Real-Time Signal Quality-Aware ECG Telemetry System for IoT-Based Health Care Monitoring // IEEE Internet Things J. 2017. vol. 4. no. 3. pp. 815–823. DOI: 10.1109/JIOT.2017.2670022.
- Al-Jaroodi J., Mohamed N. PsCPS: A Distributed Platform for Cloud and Fog Integrated Smart Cyber-Physical Systems // IEEE Access. 2018. vol. 6. pp. 41432– 41449. DOI: 10.1109/ACCESS.2018.2856509.
- 47. Pore M., Chakati V., Banerjee A., Gupta S.K.S. ContextAiDe // ACM Trans. Internet Technol. 2019. vol. 19. no. 2. pp. 1–23. DOI: 10.1145/3301444.
- Albahri O.S. et al. Fault-Tolerant mHealth Framework in the Context of IoT-Based Real-Time Wearable Health Data Sensors // IEEE Access. 2019. vol. 7, pp. 50052– 50080. DOI: 10.1109/ACCESS.2019.2910411.
- Cao K., Xu G., Zhou J., Wei T., Chen M., Hu S. QoS-Adaptive Approximate Real-Time Computation for Mobility-Aware IoT Lifetime Optimization // IEEE Trans. Comput. Des. Integr. Circuits Syst. 2019. vol. 38. no. 10. pp. 1799–1810. DOI: 10.1109/TCAD.2018.2873239.
- Dong R., She C., Hardjawana W., Li Y., Vucetic B. Deep Learning for Hybrid 5G Services in Mobile Edge Computing Systems: Learn From a Digital Twin // IEEE Trans. Wirel. Commun. 2019. vol. 18. no. 10. pp. 4692–4707. DOI: 10.1109/TWC.2019.2927312.
- Xu X., He C., Xu Z., Qi L., Wan S., Bhuiyan M.Z.A. Joint Optimization of Offloading Utility and Privacy for Edge Computing Enabled IoT // IEEE Internet Things J. 2020. vol. 7. no. 4. pp. 2622–2629. DOI: 10.1109/JIOT.2019.2944007.
- Farahani B., Barzegari M., Aliee F.S., Shaik K.A. Towards collaborative intelligent IoT eHealth: From device to fog, and cloud // Microprocess. Microsyst. 2020. vol. 72. p. 102938. DOI: 10.1016/j.micpro.2019.102938.

- Zhang T. et al. A Joint Deep Learning and Internet of Medical Things Driven Framework for Elderly Patients // IEEE Access. 2020. vol. 8. pp. 75822–75832. DOI: 10.1109/ACCESS.2020.2989143.
- Sodhro A.H., Sodhro G.H., Guizani M., Pirbhulal S., Boukerche A. AI-Enabled Reliable Channel Modeling Architecture for Fog Computing Vehicular Networks // IEEE Wirel. Commun. 2020. vol. 27. no. 2. pp. 14–21. DOI: 10.1109/MWC.001.1900311.
- Bhatia M., Sood S.K. Quantum Computing-Inspired Network Optimization for IoT Applications // IEEE Internet Things J. 2020. vol. 7. no. 6. pp. 5590–5598. DOI: 10.1109/JIOT.2020.2979887.
- Zhang J. Research on environmental monitoring trend analysis based on internet of things visualization technology // Fresenius Environ. Bull. 2020. vol. 29. no. 2. pp. 1054 – 1062. [Online]. Available at: www.scopus.com/inward/record.uri?eid=2-s2.0-85090455681&partnerID=40&md5=7c5626045c3f39320bb91185bf694295 (accessed 11.05.2023).
- Loke G. et al. Digital electronics in fibers enable fabric-based machine-learning inference // Nat. Commun. 2021. vol. 12. no. 1. p. 3317. DOI: 10.1038/s41467-021-23628-5.
- Ke R., Zhuang Y., Pu Z., Wang Y. A Smart, Efficient, and Reliable Parking Surveillance System with Edge Artificial Intelligence on IoT Devices // IEEE Trans. Intell. Transp. Syst. 2021. vol. 22. no. 8. pp. 4962–4974. DOI: 10.1109/TITS.2020.2984197.
- Wang X., Garg S., Lin H., Piran M.J., Hu J., Hossain M.S. Enabling Secure Authentication in Industrial IoT with Transfer Learning Empowered Blockchain // IEEE Trans. Ind. Informatics. 2021. vol. 17. no. 11. pp. 7725–7733. DOI: 10.1109/TII.2021.3049405.
- Lee W.J., Xia K., Denton N.L., Ribeiro B., Sutherland J.W. Development of a speed invariant deep learning model with application to condition monitoring of rotating machinery // J. Intell. Manuf. 2021. vol. 32. no. 2. pp. 393–406. DOI: 10.1007/s10845-020-01578-x.
- Ibrar M., Wang L., Muntean G.-M., Chen J., Shah N., Akbar A. IHSF: An Intelligent Solution for Improved Performance of Reliable and Time-Sensitive Flows in Hybrid SDN-Based FC IoT Systems // IEEE Internet Things J. 2021. vol. 8. no. 5. pp. 3130– 3142. DOI: 10.1109/JIOT.2020.3024560.
- 62. Brik B., Esseghir M., Merghem-Boulahia L., Snoussi H. An IoT-based deep learning approach to analyse indoor thermal comfort of disabled people // Build. Environ. 2021. vol. 203. p. 108056. DOI: 10.1016/j.buildenv.2021.108056.
- Singh P.D., Kaur R., Singh K.D., Dhiman G., Soni M. Fog-centric IoT based smart healthcare support service for monitoring and controlling an epidemic of Swine Flu virus // Informatics Med. Unlocked. 2021. vol. 26. p. 100636. DOI: 10.1016/j.imu.2021.100636.
- Bhatia M. Intelligent System of Game-Theory-Based Decision Making in Smart Sports Industry // ACM Trans. Intell. Syst. Technol. 2021. vol. 12. no. 3. pp. 1–23. DOI: 10.1145/3447986.
- Baggag A. et al. Learning Spatiotemporal Latent Factors of Traffic via Regularized Tensor Factorization: Imputing Missing Values and Forecasting // IEEE Trans. Knowl. Data Eng. 2021. vol. 33. no. 6. pp. 2573–2587. DOI: 10.1109/TKDE.2019.2954868.
- Razzaq M.A., Mahar J.A., Ahmad M., Saher N., Mehmood A., Choi G.S. Hybrid Auto-Scaled Service-Cloud-Based Predictive Workload Modeling and Analysis for Smart Campus System // IEEE Access. 2021. vol. 9. pp. 42081–42089. DOI: 10.1109/ACCESS.2021.3065597.
- 1292 Информатика и автоматизация. 2023. Том 22 № 6. ISSN 2713-3192 (печ.) ISSN 2713-3206 (онлайн) www.ia.spcras.ru

- 67. Jin H., Zhao J. Real-time energy consumption detection simulation of network node in internet of things based on artificial intelligence // Sustainable Energy Technologies and Assessments. 2021. vol. 44. no. 101004. DOI: 10.1016/j.seta.2021.101004.
- Alzamzami F., El Saddik A. Monitoring Cyber SentiHate Social Behavior During COVID-19 Pandemic in North America // IEEE Access. 2021. vol. 9. pp. 91184– 91208. DOI: 10.1109/ACCESS.2021.3088410.
- Adhikari M., Ambigavathi M., Menon V.G., Hammoudeh M. Random Forest for Data Aggregation to Monitor and Predict COVID-19 Using Edge Networks // IEEE Internet Things Mag. 2021. vol. 4. no. 2. pp. 40–44. DOI: 10.1109/IOTM.0001.2100052.
- Jurdi R., Andrews J.G., Heath R.W. Scheduling Observers Over a Shared Channel With Hard Delivery Deadlines // IEEE Trans. Commun. 2021. vol. 69. no. 1. pp. 133– 148. DOI: 10.1109/TCOMM.2020.3032172.
- Hashash O., Sharafeddine S., Dawy Z., Mohamed A., Yaacoub E. Energy-Aware Distributed Edge ML for mHealth Applications with Strict Latency Requirements // IEEE Wirel. Commun. Lett. 2021. vol. 10. no. 12. pp. 2791–2794. DOI: 10.1109/LWC.2021.3117876.
- Liang W., Li W., Feng L. Information Security Monitoring and Management Method Based on Big Data in the Internet of Things Environment // IEEE Access. 2021. vol. 9. pp. 39798–39812. DOI: 10.1109/ACCESS.2021.3064350.
- Shao S., Zhang Q., Guo S., Qi F. Task Allocation Mechanism for Cable Real-Time Online Monitoring Business Based on Edge Computing // IEEE Syst. J. 2021. vol. 15. no. 1. pp. 1344–1355. DOI: 10.1109/JSYST.2020.2988266.
- Vaidya G., Nambi A., Prabhakar T.V., Kumar V.T., Sudhakara S. Towards generating a reliable device-specific identifier for IoT devices // Pervasive Mob. Comput. 2021. vol. 76. DOI: 10.1016/j.pmcj.2021.101445.
- Rajendran S. et al. Emphasizing privacy and security of edge intelligence with machine learning for healthcare // Int. J. Intell. Comput. Cybern. 2022. vol. 15. no. 1. pp. 92–109. DOI: 10.1108/IJICC-05-2021-0099.
- Zahid N., Sodhro A.H., Kamboh U.R., Alkhayyat A., Wang L. AI-driven adaptive reliable and sustainable approach for internet of things enabled healthcare system // Math. Biosci. Eng. 2022. vol. 19. no. 4. pp. 3953–3971. DOI: 10.3934/mbe.2022182.
- Bhardwaj A. et al. Smart IoT and Machine Learning-based Framework for Water Quality Assessment and Device Component Monitoring // Environ. Sci. Pollut. Res. 2022. vol. 29. no. 30. pp. 46018–46036. DOI: 10.1007/s11356-022-19014-3.
- Alsalemi A., Himeur Y., Bensaali F., Amira A. An innovative edge-based Internet of Energy solution for promoting energy saving in buildings // Sustainable Cities and Society. 2022. vol. 78. no. 103571. DOI: 10.1016/j.scs.2021.103571.
- Khan M.A., Ghazal T.M., Lee S.-W., Rehman A. Data Fusion-Based Machine Learning Architecture for Intrusion Detection // Comput. Mater. Contin. 2022. vol. 70. no. 2. pp. 3399–3413. DOI: 10.32604/cmc.2022.020173.
- Gültekin Ö., Cinar E., Özkan K., Yazıcı A. Real-Time Fault Detection and Condition Monitoring for Industrial Autonomous Transfer Vehicles Utilizing Edge Artificial Intelligence // Sensors. 2022. vol. 229(9). no. 3208. DOI: 10.3390/s22093208.
- Wei L., Hou S., Liu Q. Clinical Care of Hyperthyroidism Using Wearable Medical Devices in a Medical IoT Scenario // Journal of Healthcare Engineering. 2022. vol. 2022. DOI: 10.1155/2022/5951326.
- Nikolov G., Kuhn M., Mcgibney A., Wenning B.-L. MABASR A Robust Wireless Interface Selection Policy for Heterogeneous Vehicular Networks // IEEE Access. 2022. vol. 10. pp. 26068–26077. DOI: 10.1109/ACCESS.2022.3156597.

- Baek J., Kaddoum G. Online Partial Offloading and Task Scheduling in SDN-Fog Networks With Deep Recurrent Reinforcement Learning // IEEE Internet Things J. 2022. vol. 9. no. 13. pp. 11578–11589. DOI: 10.1109/JIOT.2021.3130474.
- Thenmozhi R., Sakthivel P., Kulothungan K. Hybrid multi-objective-optimization algorithm for energy efficient priority-based QoS routing in IoT networks // Wireless Networks. 2022. DOI: 10.1007/s11276-021-02848-z.
- Manoharan S.N., Kumar K.M.V.M., Vadivelan N.A. Novel CNN-TLSTM Approach for Dengue Disease Identification and Prevention using IoT-Fog Cloud Architecture // Neural Processing Letters. 2022. vol. 55. no. 2. pp. 1951–1973. DOI: 10.1007/s11063-022-10971-x.
- Sithik M.M., Kumar B.M. Intelligent agent based virtual clustering and multi-context aware routing for congestion mitigation in secure RPL-IoT environment // Ad Hoc Networks. 2022. vol. 137. DOI: 10.1016/j.adhoc.2022.102972.
- Zhang Y., Wu J., Liu M., Tan A. TSN-based routing and scheduling scheme for Industrial Internet of Things in underground mining // Engineering Applications of Artificial Intelligence. 2022. vol. 115. DOI: 10.1016/j.engappai.2022.105314.
- Eroshkin I., Vojtech L., Neruda M. Resource Efficient Real-Time Reliability Model for Multi-Agent IoT Systems // IEEE Access. 2022. vol. 10. pp. 2578–2590. DOI: 10.1109/ACCESS.2021.3138931.
- Zhou S., Du Y., Chen B., Li Y., Luan X. An Intelligent IoT Sensing System for Rail Vehicle Running States Based on TinyML // IEEE Access. 2022. vol. 10. pp. 98860– 98871. DOI: 10.1109/ACCESS.2022.3206954.
- Sinha A., Das D., Udutalapally V., Mohanty S.P. iThing: Designing Next-Generation Things with Battery Health Self-Monitoring Capabilities for Sustainable IIoT // IEEE Trans. Instrum. Meas. 2022. vol. 71. pp. 1–9. DOI: 10.1109/TIM.2022.3216594.
- Manocha A., Singh R. A Novel Edge Analytics Assisted Motor Movement Recognition Framework Using Multi-Stage Convo-GRU Model // Mob. Networks Appl. 2022. vol. 27. no. 2. pp. 657–676. DOI: 10.1007/s11036-019-01321-8.
- Bollen E. et al. A database system for querying of river networks: facilitating monitoring and prediction applications // Water Supply. 2022. vol. 22. no. 3. pp. 2832–2846. DOI: 10.2166/ws.2021.433.
- Fournier A.M.V. et al. Structured Decision Making to Prioritize Regional Bird Monitoring Needs // INFORMS Journal on Applied Analytics. 2023. vol. 53(3). pp. 207–217. DOI: 10.1287/inte.2022.1154.
- Fanelli S., Pratici L., Salvatore F.P., Donelli C.C., Zangrandi A. Big data analysis for decision-making processes: challenges and opportunities for the management of health-care organizations // Management Research Review. 2023. vol. 46. no. 3. pp. 369–389. DOI: 10.1108/MRR-09-2021-0648.
- Overdal M., Haddara M., Langseth M. Exploring Public Cloud-ERP Systems' Impact on Organizational Performance // Lect. Notes Networks Syst. 2023. vol. 561. pp. 121– 137. DOI: 10.1007/978-3-031-18344-7_8.
- Reyna A., Martín C., Chen J., Soler E., Díaz M. On blockchain and its integration with IoT. Challenges and opportunities // Futur. Gener. Comput. Syst. 2018. vol. 88. pp. 173–190. DOI: 10.1016/j.future.2018.05.046.

Диван Марио Хосе — Ph.D., почетный профессор, старший архитектор решений, отделение разработки надежности вычислительных сред, Корпорация Intel. Область научных интересов: кибербезопасность, управление, масштабируемость и монетизация ИТ-сервисных платформ. Число научных публикаций — 91. mario.jose.divan.koller@intel.com; 25-я авеню, Кампус Джонс Фарм, 3, 97124, Хиллсборо, штат Орегон, Соединённые штаты Америки; р.т.: +1(503)264-0180.

Щемелинин Дмитрий Александрович — д-р техн. наук, руководитель направления, отделение разработки надежности облачных платформ, Корпорация Intel. Область научных интересов: мониторинг, управление состоянием и масштабируемость глобально распределенных вычислительных комплексов с применением методов искусственного интеллекта, сбор и анализ больших данных. Число научных публикаций — 78. dshchmel@gmail.com; 25-я авеню, Кампус Джонс Фарм, 3, 97124, Хиллсборо, штат Орегон, Соединённые штаты Америки; р.т.: +1(408)981-5963.

Карранса Маркос — главный инженер, отделение разработки надежности вычислительных сред, корпорация Intel. Область научных интересов: кибербезопасность, управление, масштабируемость и монетизация ИТ-сервисных платформ. Число научных публикаций — 32. marcos.e.carranza@intel.com; 25-я авеню, Кампус Джонс Фарм, 3, 97124, Хиллсборо, штат Орегон, Соединённые штаты Америки; р.т.: +1(503)764-7952.

Мартинес-Спессот Цезарь Игнасио — руководитель направления, отделение разработки надежности вычислительных сред, Корпорация Intel. Область научных интересов: применение цифровых технологий для улучшения бизнес-процессов с применением методов искусственного интеллекта, развитие экосистем и приложений Интернета вещей. Число научных публикаций — 31. cesar.martinez@intel.com; 25-я авеню, Кампус Джонс Фарм, 3, 97124, Хиллсборо, штат Орегон, Соединённые штаты Америки; р.т.: +1(503)7120559.

Буйневич Михаил Викторович — д-р техн. наук, профессор кафедры, кафедра прикладной математики и информационных технологий, Санкт-Петербургский университет государственной противопожарной службы МЧС России. Область научных интересов: киберустойчивость информационно-телекоммуникационных систем, методы искусственного интеллекта в обработке данных о состоянии сложных систем, подготовка специалистов в области информационной безопасности, киберполигон. Число научных публикаций — 170. bmv1958@yandex.ru; Московский проспект, 149, 196105, Санкт-Петербург, Россия; р.т.: +7(812)388-8639.

Поддержка исследований. Эта работа была полностью поддержана корпорацией Intel.